

# Working Paper No. 2018-03

# Financially-Constrained Lawyers: An Economic Theory of Legal Disputes

Claudia M. Landeo University of Alberta

Maxim Nikitin Higher School of Economics

February 2018

Copyright to papers in this working paper series rests with the authors and their assignees. Papers may be downloaded for personal use. Downloading of papers for any other activity may not be done without the written consent of the authors.

Short excerpts of these working papers may be quoted without explicit permission provided that full credit is given to the source.

The Department of Economics, the Institute for Public Economics, and the University of Alberta accept no responsibility for the accuracy or point of view represented in this work in progress.

[Forthcoming, Games and Economic Behavior]

# Financially-Constrained Lawyers: An Economic Theory of Legal Disputes<sup>\*</sup>

Claudia M. Landeo<sup>†</sup>and Maxim Nikitin<sup>‡</sup>

February 22, 2018

#### Abstract

Financial constraints reduce the lawyer's ability to file lawsuits and bring cases to trial. As a result, access to justice for victims, pretrial bargaining, and potential injurers' precaution might be affected. We study civil litigation using a model that allows for asymmetric information, financially-constrained lawyers, third-party lawyer lending, and a continuum of plaintiff's types. We contribute to the economic analysis of law by generalizing seminal models of litigation (Bebchuk, 1984, 1988; Katz, 1990), offering the first formal definition of access to justice, and presenting comprehensive social welfare analysis of relevant public policy. We provide complete equilibrium characterization and identify necessary conditions for the existence of the mixed- and pure-strategy PBE. The mixed-strategy equilibrium arises in a state of the world characterized by lawyers facing strong financial constraints. Access to justice is denied to some victims under the mixed-strategy equilibrium. We then study the social welfare effects of policies aimed at relaxing lawyers' financial constraints, and identify a necessary and sufficient condition for a welfare-enhancing effect.

KEYWORDS: Civil Litigation; Access to Justice; Social Welfare; Financially-Constrained Lawyers; Asymmetric Information; Perfect Bayesian Equilibrium; Deterrence; Lawsuits; Settlement; Litigation; Third-Party Lawyer Lending Industry; Third-Party Litigation Funding

JEL Categories: K41, C70, D82

<sup>&</sup>lt;sup>\*</sup>Claudia Landeo acknowledges financial support from the National Science Foundation (NSF Grant SES-1155761). We thank the Editor of this Journal, the Advisory Editor, and the three Referees for their insightful comments and suggestions. We also thank Emiliano Catonini, Andrew Daughety, Sergei Izmalkov, Jack Ochs, Alexei Parakhonyak, Jennifer Reinganum, Kathryn Spier, Anton Suvorov and Bart Taub for helpful discussions and comments. We are grateful for suggestions from conference and seminar participants at the 2016 NBER Summer Institute in Law and Economics, the 2016 Canadian Law and Economics Association Meeting, the 2016 Latin American and Iberian Law and Economics Association Meeting, the 2016 Latin American and Iberian Law and Economics Association Meeting, the 2016 International Conference on Game Theory, the New Economic School, and the University of Essex.

<sup>&</sup>lt;sup>†</sup>University of Alberta, Department of Economics. Henry Marshall Tory Building 7-25, Edmonton, AB T6G 2H4, Canada. landeo@ualberta.ca; corresponding author.

<sup>&</sup>lt;sup>‡</sup>International College of Economics and Finance, NRU HSE. Shabolovka 26, Moscow 119049, Russia. mnikitin@hse.ru.

# 1 Introduction

The U.S. tort system provides \$172 billion in gross compensation to plaintiffs each year. Litigation expenses, which are generally covered by personal injury lawyers on behalf of their clients, represent \$5.2 billion of this compensation (Engstrom, 2014). The average cost of taking a medical malpractice claim to trial is \$97,000 (Shepherd, 2014). Expenses on expert witnesses in the \$50,000-\$100,000 range are not uncommon (Trautner, 2009). As cases become more complex and hence, more expensive, lawyers might experience financial constraints (Engstrom, 2014; Garber, 2010). Financial constraints weaken the lawyers' ability to file lawsuits and bring cases to trial. As a result, access to justice for victims is compromised. Importantly, by affecting the pool of filed cases, lawyers' financial constraints might also influence pretrial bargaining outcomes and potential injurers' precaution. Hence, a comprehensive analysis of civil litigation should consider the financial constraints that lawyers face. Policy debate, however, has been centered only on the effects of lawyers' financial constraints on access to justice. Previous theoretical work on legal disputes has simply abstracted from lawyers' financial constraints. Our paper aims to fill these gaps. We present the first strategic model of civil litigation in an environment characterized by asymmetric information, financially-constrained lawyers, and a continuum of plaintiff's types. Our framework generalizes seminal economic models of litigation (Bebchuk, 1984, 1988; Katz, 1990).

Traditionally, financially-constrained lawyers have relied on fellow lawyers' contributions and bank loans. In the late 1990s, a third-party lawyer's lending industry emerged.<sup>1</sup> Lawyer lenders such as *Counsel Financial*, among others, started funding activities. These lawyer lending institutions specialize in providing recourse loans (non-contingent loans) to cover the expenses associated with particular legal cases (Garber, 2010). According to legal commentators, "[R]ecourse lawyer lending ... is making a significant mark ... [and has] remarkable potential for future growth" (Engstrom, 2014; p. 397).<sup>2</sup> In contrast to traditional banks, these lenders do not require lawyers'

<sup>2</sup>Due to the absence of a central repository of data, it is impossible to estimate the exact amount of loans associated with the lawyer lending industry. Engstrom (2014) provides some evidence of the scope of the industry.

<sup>&</sup>lt;sup>1</sup>The third-party litigation funding industry, and the demand for these services by plaintiffs' lawyers and clients, have been amply debated by policy analysts (see for instance the RAND Institute for Civil Justice policy reports by Garber, 2010, and McGovern, et al., 2010), studied by legal scholars (see for instance, Molot, 2010, 2009; Rodak, 20016; Steinitz, 2012; and the fifteen articles included in the *DePaul Law Review* symposium volume on litigation and law firm finance, 2014 (see Landsman et al. in the References section)), and discussed in the media (see for instance, *The Economist*, April 6th 2013; the article by Glater published at *The New York Times*, June 2 2009; the article by Frankel published at *The American Lawyer*, February 13 2006; and, the article by Carter published at The American Bar Association's *ABA Journal*, October 2004). However, there are no previous formal models of civil litigation involving financially-constrained lawyers and lawyer's lending institutions.

personal assets as collaterals. Instead, the loans are secured by the law firm's assets, including future fees (Garber, 2010).<sup>3</sup> The loans involve significantly larger sums and the interest charged is higher than the traditional bank's interest (around 15-20 percent per year). Our theoretical framework also captures the role of the lawyer lending industry on legal disputes.

We model the interaction between a defendant, a plaintiff, and a plaintiff's lawyer as a sequential game of incomplete information. Our framework allows for asymmetric information, financially-constrained lawyers, third-party lawyer lending, and a continuum of plaintiff's types. The source of information asymmetry is the damage level of the plaintiff's case, which is unknown by the defendant. Our original framework depicts legal disputes from cradle to grave. Specifically, our model allows for endogenous care-taking (precaution), endogenous filing, and endogenous out-of-court settlement decisions. We model the lawyers' financial constraints by incorporating real-world characteristics of the third-party lawyer lending industry. We present the first formal definition of "Access to Justice" and incorporate this concept in the social welfare analysis of relevant public policy.

We provide complete characterization of the two mutually-exclusive perfect Bayesian equilibria (PBE): Mixed- and pure-strategy equilibria. In the mixed-strategy equilibrium, a lawyer with a low-damage case mixes between accepting the case and filing a lawsuit and not accepting the case, and a lawyer with a high-damage case always accepts the case and files a lawsuit; the defendant mixes between making a zero offer and a positive offer. In the pure-strategy equilibrium, lawyers with all types of cases always accept the cases and file lawsuits; and, the defendant always makes a positive offer. Across equilibria, accidents and pre-trial bargaining disagreement are observed. Access to justice is denied to some victims only under the mixed-strategy equilibrium.

We identify necessary conditions for the existence of each PBE. In particular, our analysis demonstrates that the pure-strategy equilibrium arises in a state of the world characterized by lawyers facing mild financial constraints. In contrast, the mixed-strategy equilibrium arises in a state of the world characterized by lawyers facing strong financial constraints. The intuition is as follows. When the level of lawyers' financial constraints is high, then low-damage plaintiffs (i.e., plaintiffs with cases that cannot proceed to trial due to the limited financial resources of

An important lawyer lender, "*Counsel Financial*, apparently had 'more than \$200 million' in loans outstanding as of 2010" (p. 397). Lawyer lenders operate in almost all U.S. states. For instance, *Advocate Capital* funds law firms in forty states. Public records for the state of New York indicate that 250 law firms borrowed from lawyer lending providers during the period 2000–2010.

<sup>&</sup>lt;sup>3</sup>Loans are secured by the estimated value of a firm's total portfolio of cases. Lenders generally require access to the firm's entire docket of cases and assets.

their lawyers and/or high litigation costs) are relatively common. The defendant will reduce his expected litigation loss by mixing between a zero offer and a positive offer (instead of making a unique positive offer) because low-damage plaintiffs who receive a zero offer will drop their cases. In contrast, a unique positive offer will be accepted by all low-damage plaintiffs.

We then use our model to study the effects of policies aimed at relaxing lawyers' financial constraints by lowering the costs associated with legal disputes. The interest charged to lawyers by third party lenders and the costs associated with expert witnesses' fees hired by plaintiff's lawyers are examples of such costs.<sup>4</sup> Given that these policies do not affect equilibrium strategies and outcomes under the pure-strategy PBE, we focus our analysis on the mixed-strategy PBE. Our findings suggest that cost-reducing policies increase filing of low-damage cases. The uninformed defendant reacts by reducing his positive out-of-court settlement offer, and the likelihood of trial increases. Higher expected litigation costs for the defendant are observed, which increase his care-taking incentives, and lower the probability of an accident. Our findings also suggest that these policies improve access to justice by allowing additional victims to file lawsuits and by reducing the number of victims forced to drop their lawsuits.

Finally, we present social welfare analysis of cost-reducing policies and identify a necessary and sufficient condition for a positive welfare effect of these policies. We demonstrate that, in a state of the world characterized by strong society's concerns about preserving citizens' right of access to justice, a relaxation of the lawyers' financial constraints always increase social welfare. The intuition is as follows. As a result of relaxing the lawyers' financial constraints, many additional victims will get access to justice. In addition, potential injurers, initially underdeterred, will have more incentives to spend resources on accident prevention to avoid costly litigation. This will contribute to the alignment of privately- and socially-optimal probabilities of accidents. These positive welfare effects more than offset the higher social costs associated with more filing and litigation. Hence, social welfare will be improved.

Strong policy implications are derived from our work. Our complete social welfare analysis of cost-reducing policies provides clear policy recommendations. Importantly, this analysis applies to other policies aimed at alleviating lawyers' constraints. For instance, legal commentators are currently proposing the implementation of a cost-shifting policy that allows lawyers to transfer part of the financial costs of loans to their clients (Engstrom, 2014). Proponents claim that, by

<sup>&</sup>lt;sup>4</sup>Policies devoted to strengthening competition in the lawyer lending industry might reduce the costs of thirdparty loans. Similarly, policies designed to increase efficiency of legal procedures and reduce unpredictability of the legal system by capping punitive damages (Landeo et al., 2007b) might lower the plaintiffs' costs associated with litigation.

improving access to justice, this policy will be welfare-enhancing. We contribute to this debate by underscoring the importance of assessing public policies in terms of their overall effect on social welfare, and by identifying a necessary and sufficient condition for a positive welfare effect of a reduction in lawyers' constraints. Our analysis suggests that cost-shifting policies, which also alleviate lawyers' constraints, should be implemented when preserving access to justice for victims is a fundamental goal of society.

Several important methodological contributions to the economic analysis of law are derived from our work. We generalize seminal models of civil litigation by endogenizing the decisions associated with every step of a legal dispute. Our environment allows for asymmetric information, financially-constrained lawyers, third-party lawyer lending, and a continuum of plaintiff's types. Importantly, we offer the first formal characterization of Access to Justice, and provide a comprehensive welfare analysis of relevant public policies.

The closest to our work are Bebchuk (1984, 1988) and Katz (1990). Bebchuk (1984) studies settlement and litigation. The paper focuses on a pure-strategy PBE with a unique positive settlement offer. This equilibrium resembles the equilibrium strategies of the litigation stage that correspond to our pure-strategy PBE. Bebchuk (1988) extends Bebchuk (1984) by including a filing stage and allowing for frivolous lawsuits. In this environment, filing costs are equal to zero. Then, all potential plaintiffs have an incentive to file a lawsuit. The paper focuses on a purestrategy PBE where all potential plaintiffs file a lawsuit, and all plaintiffs receive a positive outof-court settlement offer. An important limitation of Bebchuk (1984, 1988) is that, unrealistically, all victims get access to justice in these environments. Importantly, a care-taking stage is not included in these model. Hence, these frameworks do not allow for a complete welfare analysis of relevant public policies.

Katz (1990) extends Bebchuk (1988) by relaxing the assumption of costless filing. Binary-type (meritorious and frivolous plaintiff's types) and continuum-type models are studied. Our work differs from Katz (1990) in several fundamental aspects. First, our model generalizes Katz (1990) by including a care-taking stage, by introducing the plaintiff's lawyer as a third player, and by allowing for lawyer's financial constraints and third-party lawyer lending. The introduction of a financially-constrained plaintiff's lawyer as a third player and third-party lawyer lending permits us to study the effect of public policies aimed at relaxing lawyers' financial constraints. Second, although the sequence of moves of the continuum-type model analyzed in Katz (1990) is similar to the sequence of moves of the filing and litigation stages in our model, the focus of Katz's (1990) analysis of the continuum-type model is on the pure-strategy PBE where a unique positive out-of-court settlement offer is always made. Hence, a comprehensive analysis of access to justice cannot

be performed. In contrast to this paper, we formally characterize access to justice, and perform comparative-statics analysis of the effect of cost-reducing policies on access to justice. Third, we provide a comprehensive social welfare analysis of the effect of policies aimed at alleviating lawyers' financial constraints and identify a necessary and sufficient condition for a welfare-enhancing effect of these policies. Importantly, by endogenizing the care-taking decisions, our model permits us to assess not only the direct welfare effect of public policies that operates through the filing and litigation decisions but also the indirect welfare effect that operates through the defendant's caretaking decisions. Hence, the first and most important contribution of our paper is to present an original model of civil litigation that allows for a comprehensive social welfare analysis of relevant public policies.<sup>5</sup>

Our paper also contributes to the literature on third-party litigation funding. As discussed in Garber (2010), third-party litigation funding includes three types of financing: Recourse loans to plaintiffs' lawyers, non-recourse loans to individual plaintiffs, and investments in commercial lawsuits (business against business lawsuits). Previous theoretical work on third-party litigation funding has been focused on the plaintiff's lending segment of the industry. Daughety and Reinganum (2014) study the design of third-party loans to plaintiffs using dynamic incomplete information models with the plaintiff as the first mover.<sup>6</sup> Their findings suggest that the third-party institution affects only pre-trial bargaining outcomes.<sup>7</sup> Analysis of the effects of the third-party institution on filing is not included. We provide several contributions to the literature on thirdparty litigation funding. First, our work contributes to a better understanding of the third-party litigation funding institution by presenting the first formal analysis of legal disputes in an environment that allows for third-party lawyer lending institutions. Second, our work incorporates care-taking and filing decisions into the analysis, and demonstrates that changes in the features of the third-party litigation institution might affect not only settlement but also care-taking in-

<sup>6</sup>See also Deffains and Desrieux (2015), Avraham and Wickelgren (2014), Demougin and Maultzsch (2014) and Hylton (2012). These papers are focused on the plaintiff's lending segment of the third-party litigation funding industry, and only study the litigation stage.

<sup>7</sup>In this framework, the defendant's care taking incentives are not affected by the third-party institution.

<sup>&</sup>lt;sup>5</sup>For additional seminal theoretical work on civil litigation, see Shavell (1982), Png (1983), Rosenberg and Shavell (1985), Reinganum and Wilde (1986) and Nalebuff (1987). These models are focused on the litigation stage. See Landeo et al. (2007a, b; 2006) and Png (1987) for seminal civil litigation models that allow for endogenous care-taking decisions. In contrast to our work, all these models do not allow for constraints at the litigation stage (i.e., they assume that the party who makes the decision at the litigation stage has sufficient funds to cover the litigation costs) and do not study access to justice. See Landeo (forthcoming) and Daughety and Reinganum (2012) for recent surveys of theoretical and experimental work on civil litigation.

centives and filing decisions. In particular, we show that in a state of the world characterized by strong society's concern about access to justice by victims, policies aimed at reducing the interest rate charged on third-party loans induce more low-damage cases to be filed, increase the likelihood of trial, increase the defendant's care-taking incentives, and increase social welfare.

The rest of the article is organized as follows. Section Two presents the setup of the model and provides the formal definition of Access to Justice. Section Three outlines the equilibrium analysis. Section Four discusses the effect of a cost-reducing policy on equilibrium outcomes and access to justice. Section Five provides formal characterization of the social welfare loss function and presents an analysis of deterrence. Section Six presents the social welfare analysis of a costreducing policy. Section Seven provides concluding remarks. The Appendix includes formal proofs of the mixed- and pure-strategy PBE of our general model, and proofs related to deterrence and the social welfare analysis of a cost-reducing policy.

# 2 Model Setup

This section describes the game stages, the "Lawyer's Constraint" component, and the "No-Access to Justice" component. It also introduces the notation.

# 2.1 Game Stages

We model the interaction between a potential defendant,<sup>8</sup> a potential plaintiff, and a potential plaintiff's lawyer as a sequential game of incomplete information. The source of information asymmetry is the damage level of the potential plaintiff's case A, which is unknown by the defendant. The stages of the game are as follows.

### **Care-Taking Stage**

In the first stage, the potential defendant decides his level of care, which determines the probability of accident  $\lambda$ . The cost of care is denoted by  $K(\lambda)$ . We assume that all potential defendants have the same cost of care, which is common knowledge. We also assume that  $K(\lambda)$  is a continuous and differentiable function defined on the interval [0, 1] with with K(1) = 0,  $\frac{\partial K(\lambda)}{\partial \lambda} < 0$ ,  $\lim_{\lambda \to 0^+} \frac{\partial K(\lambda)}{\partial \lambda} =$  $-\infty$ ,  $\lim_{\lambda \to 1^-} \frac{\partial K(\lambda)}{\partial \lambda} = 0$ ; and, that  $\frac{\partial K(\lambda)}{\partial \lambda}$  is a continuous and differentiable function with  $\frac{\partial K^2(\lambda)}{\partial \lambda^2} > 0$ . The potential defendant's optimal level of care, i.e., the optimal  $\lambda$ , is the one that minimizes

<sup>&</sup>lt;sup>8</sup>We use the terms "potential defendant," "potential injurer," and "defendant" interchangeably.

the defendant's expected total loss  $L_D(\lambda) = K(\lambda) + \lambda l_D$ , where  $l_D$  is the defendant's expected litigation loss. We take the expected litigation loss as parametric in order to describe  $L_D$ , but ultimately  $l_D$  will be derived as the continuation value of the litigation stage, and hence, it will reflect the equilibrium outcomes at the litigation stage. We assume that accident occurrence is common knowledge. If an accident occurs, we assume that there is a mass 1 of potential plaintiffs. Nature determines the damage level A and informs this to the potential plaintiff only. We assume that the damage levels A are distributed according to the probability density function g(A) and the cumulative distribution function G(A), with support  $(0, \overline{A}]$ . We also assume that g(A) is a continuous and differentiable function. The functions are common knowledge.<sup>9</sup>

#### Filing Stage

If an accident occurs, the second stage starts. We assume that the mass of potential lawyers is greater than or equal to 1, i.e., it is sufficiently high to serve all potential plaintiffs. The potential plaintiff and his potential lawyer meet.<sup>10</sup> The lawyer, who perfectly observes the potential plaintiff's type (damage level), decides whether to take the case and file a lawsuit. We denote the mass of filed cases as  $\zeta$ . The lawyer is hired under a contingency-fee compensation. Under this scheme, the lawyer receives gross payment equal to a share  $\gamma$  of the plaintiff's gross recovery (award at trial or out-of-court settlement amount).<sup>11</sup> The lawyer's share  $\gamma$  is an exogenous constant, known by all parties.<sup>12</sup> The lawyer's costs of taking the case and filing a lawsuit are denoted by f, common knowledge. Following empirical regularities, we assume that the plaintiff is financially constrained. Then, the plaintiff's lawyer pays f.

### Litigation Stage

If a lawsuit is filed, the third stage starts. The uninformed defendant makes a take-it-or-leave-it out-of-court settlement offer to the plaintiff. Zero and positive offers are possible. S > 0 denotes a positive offer.  $\beta$  denotes the probability that a defendant makes a zero offer. The plaintiff then

<sup>&</sup>lt;sup>9</sup>For simplicity, we abstract from non-meritorious potential plaintiffs (i.e., frivolous cases). Our equilibrium analysis and comparative statics are robust to the presence of the threat of frivolous lawsuits.

<sup>&</sup>lt;sup>10</sup>For simplicity, our framework abstracts from searching costs.

<sup>&</sup>lt;sup>11</sup>A typical contingency fee involves a lawyer's share equal to 33-45% of any eventual recovery by his client. See *Rule 1.5(d) of the Model Rules of Professional Conduct* and statutory rules enacted by U.S. states. See Miceli (1994) for seminal formal work on contingency fees.

<sup>&</sup>lt;sup>12</sup>Our main qualitative findings hold in an environment that allows for different values of  $\gamma$  in case of trial and out-of-court settlement. See Emons (2007).

decides whether to accept or reject the defendant's offer. If an offer S > 0 is accepted by the plaintiff, then the defendant transfers the settlement amount to the plaintiff, and the game ends. The plaintiff gets  $(1 - \gamma)S$  and his lawyer gets a net payoff of  $\gamma S - f$ . Acceptance of a zero offer implies that the plaintiff drops the case. If an out-of-court settlement offer is rejected, the case proceeds to costly trial. We denote the mass of cases that proceed to trial as  $\rho$ . Both litigants incur litigation costs:  $C_P$  denotes the plaintiff's litigation cost, which is paid by his lawyer; and,  $C_D$ denotes the defendant's litigation cost. We assume that the court perfectly observes the plaintiff's type A. When a case goes to trial, the court orders the defendant to pay A to the plaintiff.<sup>13</sup> The plaintiff gets  $(1 - \gamma)A$  at trial and his lawyer gets a net payoff of  $\gamma A - C_P - f$ .

# 2.2 Lawyer's Constraint Component

This section describes the lawyer's constraint component of our model. We denote the amount of the lawyer's own funds as x > f. We assume that the lawyer is financially constrained. His own funds x are insufficient to bring a case to trial, i.e.,  $x < f + C_P$ . We also assume that there are available third-party lawyer lenders that can lend money to the lawyer to allow him to bring a case to trial, and that the third-party lawyer lending industry is not perfectly competitive.<sup>14</sup> To be able to bring a case to trial, the financially-constrained lawyer needs to borrow  $C_P + f - x$ at a net interest rate r. We denote  $\tilde{A}$  as the damage threshold at which the lawyer's net payoff when the case proceed to trial and his net payoff when the case is dropped (i.e., when a zero offer is accepted) are the same:  $\gamma \tilde{A} - f - C_P - (C_P + f - x)r = -f$ . Then, cases with  $A < \tilde{A}$  will not proceed to trial. Hence,  $\tilde{A}$  represents the lawyer's constraint component, i.e., the lawyer's inability to bring a case to trial.

**Definition 1.** The lawyer's constraint component  $\tilde{A}$  is defined as follows.

$$\tilde{A} \equiv \frac{C_P}{\gamma} + \frac{(C_P + f - x)r}{\gamma}.$$

The lawyer's constraint component encompasses two terms: (1) The litigation-cost constraint  $LC \equiv \frac{C_P}{\gamma}$ , which denotes the lawyer's constraint associated with costly litigation; and, (2) the

<sup>&</sup>lt;sup>13</sup>We assume that the court applies a strict liability rule. Under this rule, the injurer has to bear the costs of the accident regardless of the extent of his precaution. Strict liability is common in tort law cases. Then, this assumption is empirically relevant. Our tractable model can be easily extended to accommodate the court's application of a negligence rule in an environment where the defendant's level of care is common knowledge.

<sup>&</sup>lt;sup>14</sup>The specialized nature of the service provided by these lenders might act as a barrier to entry. As a result, lenders might have market power.

financial constraint  $FC \equiv \frac{(C_P + f - x)r}{\gamma}$ , which denotes the lawyer's constraint associated with the financial costs of third-party loans (due to the lawyer's limited own funds). Then,  $\tilde{A} \equiv LC + FC$ .<sup>15</sup> It is worth noting that our model provides a more accurate characterization of the constraints confronted by litigants in real-world settings than previous models of litigation. In fact, seminal models of litigation abstract from the role of the plaintiff's lawyer, and implicitly assume that the plaintiff's own funds are sufficient to cover the costs associated with litigation. Then, in those models, the plaintiff's inability to bring a case to trial is determined by  $C_P$  only. Importantly, our approach to incorporate financially-constrained lawyers to a model of litigation is aligned with empirical regularities. In fact, in his descriptive analysis of the lawyer-lending industry, Garber (2010) states: "Plaintiffs' lawyers pursuing personal-injury claims typically work on a contingent-fee basis, may have insufficient hourly work to provide steady streams of revenue, and incur out-of-pocket expenses to pursue their clients' claims" (p. 23). Our model captures these features.

In our framework, potential plaintiffs with  $A \geq \tilde{A}$  are called "high-damage cases," and potential plaintiffs with  $0 < A < \tilde{A}$  are called "low-damage cases." Remember that a measure 1 represents the potential plaintiffs. Then,  $G(\tilde{A})$  represents the mass of low-damage potential plaintiffs, and  $1 - G(\tilde{A})$  represents the mass of high-damage potential plaintiffs. We denote the mass of low-damage cases that are filed as  $\nu$ . We assume that the lawyer's constraint component  $\tilde{A}$  is common knowledge. In particular, the plaintiff knows that the lawyer's constraint allows only high-damage cases ( $A \geq \tilde{A}$ ) to proceed to trial, and takes this information into consideration when deciding whether to accept a settlement offer. Under this assumption, the plaintiff holds the settlement decision authority but her decision is conditioned by her lawyer's constraints. Then, this assumption might be interpreted as a partial delegation of settlement authority to the lawyer. According to the *Restatement (Third) of Law Governing Lawyers* (1988, Sec. 33.1), the plaintiff can validly make partial or full delegation to his lawyer. Importantly, delegation of settlement authority is commonly observed in real-world settings (Miller, 1987).<sup>16</sup>

<sup>&</sup>lt;sup>15</sup>Even if the third-party litigation funding industry is perfectly competitive (r = 0) or if the lawyer's own funds are sufficient to cover the costs associated with litigation  $(x \ge C_P + f)$ , when  $A < \tilde{A}$ , the lawyer will still face a litigation-cost constraint. We thank a referee for pointing this out.

<sup>&</sup>lt;sup>16</sup>See Dana and Spier (1990) for a litigation model involving full delegation of authority to the lawyer regarding the decision to drop the case. See Choi (2003) for a litigation model involving plaintiff's endogenous decision of delegation of settlement authority to the lawyer.

# 2.3 No-Access to Justice Component

This section formally defines the "No-Access to Justice" component of our model. The "No-Access to Justice" component  $\eta$  represents the inability of meritorious (potential) plaintiffs to get access to justice. To the best of our knowledge, ours represents the first formal definition of "No-Access to Justice."

**Definition 2.** No-Access to Justice  $\eta$  is defined as the sum of two terms.

- (1) The mass of meritorious cases that are not filed.
- (2) The mass of meritorious cases that receive a zero offer and cannot proceed to trial.

Intuitively, the first term represents the inability of lawyers with meritorious cases to file a lawsuit. The second term represents the inability of meritorious plaintiffs to get compensation for the inflicted injury through a positive out-of-court settlement transfer or an award at trial. The specific elements included in  $\eta$  depend on the equilibrium outcomes.

# 3 Equilibrium Analysis

Our analysis demonstrates that the lawyers' financial constraints affect the equilibrium structure and permeates the decisions of all the parties involved in a legal dispute. We completely characterize the two-mutually exclusive PBE of the model: Pure- and mixed-strategy PBE. In the pure-strategy PBE, the defendant always make a positive offer and all victims file lawsuits. In the mixed-strategy equilibrium, the defendant randomizes between a positive and a zero offer, and not all victims are able to file a lawsuit. We show that, across equilibria, accidents occur and bargaining disagreement at the litigation stage is observed. Access to justice is compromised only in case of the mixed-strategy PBE. In real-world settings, access to justice is denied to some victims. Then, the mixed-strategy PBE provides a better description of the actual state of affairs in civil litigation.

We provide necessary conditions for the existence of the two PBE. The pure-strategy PBE arises in a state of the world characterized by lawyers facing mild financial constraints; and the mixed-strategy PBE arises in a state of the world characterized by lawyers facing strong financial constraints. The intuition is as follows. When the level of lawyers' financial constraints is high, then low-damage plaintiffs (i.e., plaintiffs with cases that cannot proceed to trial due to the limited financial resources of their lawyers and/or high litigation costs) are relatively common and high-damage plaintiffs are relatively rare. The defendant will reduce his expected litigation loss by

mixing between a zero offer and a positive offer because low-damage plaintiffs who receive a zero offer will drop their cases and only high-damage plaintiffs will reject a zero offer and proceed to trial; a unique positive offer, in contrast, will be accepted by all low-damage plaintiffs. Hence, when the level of lawyers' financial constraints is high, then the defendant will have a strong incentive to mix between a zero offer and a positive offer. In contrast to previous theoretical models of litigation (Bebchuk, 1984, 1988; Katz, 1990), our work shows that the plaintiff's litigation costs influence the lawyer's constraints, and hence, might affect not only the behavior of the litigants at the litigation stage but also the structure of the equilibrium.

The next sections outline the main steps in the construction of the mixed- and pure-strategy PBE. Formal proofs are presented in the Appendix.

# 3.1 Mixed-Strategy PBE

In a state of the world characterized by strong lawyers' financial constraints, the mixed-strategy PBE emerges. In the mixed-strategy equilibrium, the defendant mixes between making a zero offer and a positive offer; and, lawyers with high-damage cases always file a lawsuit and lawyers with low-damage cases mix between accepting the case and filing a lawsuit and not accepting the case. Due to asymmetric information, some high-damage plaintiffs receive an insufficiently high offer (i.e., an offer below their actual harm from the accident) and proceed to costly trial; and, some low-damage plaintiffs receive a generous offer (an offer above their actual harm from the accident) and settle out-of-court, and some low-damage plaintiffs receive a zero offer and need to drop their cases. Given that only a subset of low-damage cases are filed, and some low-damage cases are dropped, then some low-damage victims do not get access to justice.

# **Equilibrium Strategies**

#### Step 1: Potential Composition of the Set of Equilibrium Offers

First, we analyze the potential composition of the set of equilibrium offers and demonstrate that this set must include a zero offer and at least one positive offer  $S \in [\tilde{A}, \bar{A}]$  (see Lemma 1 and Claim 1 in the Appendix). We start with the elimination of the strictly dominated offers  $S > \bar{A}$ and offers  $S \in (0, \tilde{A})$ . We then demonstrate that a zero offer must be made in equilibrium. If a zero offer is not made in equilibrium, then a positive offer  $S \in [\tilde{A}, \bar{A}]$  must be made. Assume that the filing costs are low enough  $(f < \gamma \tilde{A})$ . Then, all low- and high-damage cases will be filed. As shown in the proof of Lemma 1, the defendant's expected litigation loss from making the positive offer will be greater than his expected litigation loss from making a zero offer. Contradiction follows. Hence, a zero offer must be made in equilibrium, and the probability of making a zero offer should be greater than zero ( $\beta > 0$ ).

Finally, we show that a zero offer cannot be the only equilibrium offer. If the defendant always makes a zero offer, then only high-damage cases with types  $A \ge \hat{A} > \tilde{A}$  will be filed.<sup>17</sup> These types of plaintiffs will always reject a zero offer. Then, the defendant will be better off by deviating from a zero offer to a positive offer slightly higher than  $\hat{A}$ . Hence, a zero offer cannot be the only equilibrium offer. Given our previous findings, in addition to a zero offer, there must be at least one positive offer  $S \in [\tilde{A}, \bar{A}]$  in the set of equilibrium offers.

#### Step 2: Potential Composition of the Equilibrium Mass of Filed Cases $\zeta$

Next, we study the potential composition of the equilibrium mass of filed cases  $\zeta$ , and show that at least some low-damage cases and all high-damage cases must be filed in equilibrium (see Lemma 2 in the Appendix). We start with the evaluation of filing of low-damage cases. If no low-damage cases are filed in equilibrium (i.e., if  $\nu = 0$ ), then a mixed strategy involving a zero offer and at least one positive offer  $S \in [\tilde{A}, \bar{A}]$  is not an equilibrium strategy. As we show in the proof of Lemma 2, the defendant will be better off by deviating from the mixed-strategy involving a zero offer and a positive offer  $S \in [\tilde{A}, \bar{A}]$  to a pure-strategy involving a positive offer greater than  $\tilde{A}$ . The intuition is as follows. If low-damage cases are not filed, then a zero offer does not provide any benefits for the defendant: A zero offer offer will always be rejected by high-damages plaintiffs, and hence, no cases will be dropped. The defendant will economize on litigation costs by deviating to a pure strategy with a positive offer greater than  $\tilde{A}$ . We conclude that at least some low-damage cases must be filed.

We proceed with the study of filing of high-damage cases. Given that at least some low-damage cases are filed in equilibrium, it must be the case that at least some lawyers with low-damage cases will get non-negative expected payoffs:  $\gamma[\beta(0) + (1 - \beta)S)] - f \ge 0$ , where S and  $\beta$  represent a positive offer and the probability of getting a zero offer, respectively. Remember also that the equilibrium probability that the defendant makes a zero offer is positive ( $\beta > 0$ ). Consider now the case of a lawyer with a high-damage case  $A \ge \tilde{A}$ . The expected payoff for the lawyer is  $\beta[\gamma A - C_P - (C_P + f - x)r] + (1 - \beta)\gamma S - f$ . This expression is non-negative for  $A = \tilde{A}$  and positive for  $A > \tilde{A}$ . As a consequence, all high-damage cases will be filed. Hence, the equilibrium

<sup>&</sup>lt;sup>17</sup>Note that  $\hat{A} \equiv \frac{C_P + f + (C_P + f - x)r}{\gamma}$  refers to the damage threshold at which the lawyer's net payoff is zero when a high-damage case is solved at trial.

mass of cases that are filed  $\zeta = \nu + \int_{\tilde{A}}^{\tilde{A}} g(A) dA$ , where  $\nu > 0$  represents the mass of low-damage cases that are filed.

#### Step 3: Equilibrium S and $\nu$

Now, we are in the position to analyze the equilibrium positive settlement offer S and the equilibrium mass of low-damage cases that are filed  $\nu$  (see Lemmas 3 and 4 in the Appendix). We will show that equilibrium S and  $\nu$  exist and are unique.<sup>18</sup> Remember that lawyers with high-damage cases always file lawsuits, and some lawyers with low-damage cases also file lawsuits; and, that there are at least two equilibrium offers, a zero offer and a positive  $S \in [\tilde{A}, \bar{A}]$ . This implies that both a zero offer and the positive offer S must minimize the defendant's expected litigation loss  $l_D$ , and that the defendant must be indifferent between these two offers. The defendant's indifference condition is given by:

$$\nu S + \int_{\tilde{A}}^{S} Sg(A)dA + \int_{S}^{\tilde{A}} (A + C_D)g(A)dA = \int_{\tilde{A}}^{\tilde{A}} (A + C_D)g(A)dA.$$
(1)

The left- and right-hand sides of equation (1) represent the defendant's expected litigation loss  $l_D$  from making an offer S > 0 and a zero offer, respectively. Then, the first-order optimality condition is:

$$\frac{\partial}{\partial S} \left[ \nu S + \int_{\tilde{A}}^{S} Sg(A) dA + \int_{S}^{\tilde{A}} (A + C_D)g(A) dA \right] = 0.$$

This last equation simplifies to:

$$\nu + G(S) - G(A) = C_D g(S).$$
(2)

Equation (2) indicates that the marginal cost of raising S (represented by the mass of plaintiffs who accept S; left-hand side of the equation) equals the marginal benefit of raising S (represented by the savings in litigation cost  $C_D$  as fewer cases go to trial; right-hand side of the equation). As demonstrated in Lemma 3, the system of equations (1)–(2) has a unique solution, S and  $\nu$ . When,  $g(A) - C_D \frac{\partial g(A)}{\partial A} > 0$  for any  $A \in [\tilde{A}, \bar{A}]$ , the second-order optimality condition is satisfied. Finally, when  $\int_{\tilde{A}}^{\tilde{A}} Ag(A) dA + C_D [1 - G(\tilde{A}) - \bar{A}g(\bar{A})] > 0$ , the optimal positive offer S is interior.

Figure 1 presents a graphical representation of the optimal decision of the defendant at the litigation stage. The defendant's expected litigation loss function  $l_D$  is constructed by keeping the

<sup>&</sup>lt;sup>18</sup>To simplify notation, we also refer to the *equilibrium* positive offer and mass of low-damage cases that are filed as S and  $\nu$ , respectively, in this section and the rest of the paper.



Figure 1: Defendant's Equilibrium Mixed-Strategy at the Litigation Stage

equilibrium mass of filed cases  $\zeta = \nu + \int_{\tilde{A}}^{\bar{A}} g(A) dA$  constant.

$$l_D = \begin{cases} \int_{\tilde{A}}^{\tilde{A}} (A+C_D)g(A)dA & \text{if zero offer} \\ \nu S + \int_{\tilde{A}}^{\tilde{A}} (A+C_D)g(A)dA & \text{if } S \in (0,\tilde{A}) \\ \nu S + \int_{\tilde{A}}^{S} Sg(A)dA + \int_{S}^{\tilde{A}} (A+C_D)g(A)dA & \text{if } S \in [\tilde{A}, \bar{A}] \end{cases}$$

When  $S \in (0, \tilde{A})$ ,  $l_D$  is linear and upward-sloping. When  $S \in [\tilde{A}, \bar{A}]$ ,  $l_D$  is strictly convex and achieves a unique interior minimum. The defendant's expected loss function  $l_D$  attains the same value at a zero offer and at the optimal positive offer S. Hence, in equilibrium, the defendant mixes between a zero offer and a positive (interior) offer  $S \in (\tilde{A}, \bar{A})$ .

### Step 4: Composition of Equilibrium $\nu$ and $\beta$

Next, we study the composition of equilibrium mass of low-damage cases that are filed  $\nu$  and the equilibrium probability that the defendant makes a zero offer  $\beta$  (see Claim 2 and Lemma 5 in the Appendix). Remember that at least some low-damages cases are filed in equilibrium. Claim 2 in the Appendix demonstrates that the total mass of low-damage cases  $G(\tilde{A})$  is greater than the equilibrium mass of low-damage cases that are filed  $\nu$ . Hence, not all low-damage cases are filed in equilibrium. Intuitively, the defendant's randomization between a zero offer and a positive offer S induces a lawyer with a low-damage case to randomize between filing and not filing a lawsuit. Although low-damage cases cannot proceed to trial, some of these cases might still receive a positive out-of-court settlement offer due to asymmetric information.

A lawyer with an average low-damage case who randomizes between filing and not filing a lawsuit must be indifferent between these two strategies.<sup>19</sup> In other words, his expected net payoff

<sup>&</sup>lt;sup>19</sup>In principle, the probability of filing for a low-damage plaintiff may depend on the specific A. Then, the expression "average low-damage client" is used here.

from filing and not filing should be the same:  $0 = \gamma[\beta(0) + (1 - \beta)S] - f$ . Using the lawyer's indifference condition, we can compute the equilibrium probability that a defendant makes a zero offer  $\beta$ :  $\beta = 1 - \frac{f}{\gamma S}$ .

### Step 5: Equilibrium $\lambda$

Finally, we assess the optimal decision of the defendant at the care-taking stage. Given the equilibrium outcomes at the filing and litigation stages, the defendant's optimal probability of an accident is  $\lambda_D = \arg \min\{K(\lambda) + \lambda l_D\}$ , where  $l_D = \int_{\tilde{A}}^{\tilde{A}} (A + C_D)g(A)dA$  represents the defendant's expected litigation loss in equilibrium. The proof of Lemma 6 in the Appendix demonstrates that for any positive value of  $l_D$ , the function  $K(\lambda) + \lambda l_D$  has a unique interior minimum  $\lambda_D \in (0, 1)$ .

Proposition 1 characterizes the mixed-strategy PBE under conditions (3)-(6).

$$f < \gamma \tilde{A}.$$
 (3)

$$\min_{\tau \in [\tilde{A}, \bar{A}]} \left\{ [G(\tau)]\tau - \int_{\tilde{A}}^{\tau} (A + C_D)g(A)dA \right\} > 0.$$

$$\tag{4}$$

For any  $A \in [\tilde{A}, \bar{A}]$ ,

$$g(A) - C_D \frac{\partial g(A)}{\partial A} > 0.$$
(5)

$$\int_{\tilde{A}}^{A} Ag(A)dA + C_{D}[1 - G(\tilde{A}) - \bar{A}g(\bar{A})] > 0.$$
(6)

Condition (3) implies that the lawyer's net payoff  $(\gamma \tilde{A} - f)$  is positive when an offer greater than or equal to  $\tilde{A}$  is made in equilibrium. Then, this is a sufficient condition for low-damage cases to be filed if such an offer is made in equilibrium.

Condition (4) implies that, for any positive offer  $\tau \in [\tilde{A}, \bar{A}]$ , the defendant's expected loss from making this positive offer,  $G(\tau)\tau + \int_{\tau}^{\bar{A}} (A + C_D)g(A)dA$ , is strictly greater than his expected loss from making a zero offer,  $\int_{\tilde{A}}^{\bar{A}} (A + C_D)g(A)dA$ . This condition ensures that a zero offer is in the set of equilibrium offers, and hence, it is a necessary condition for the existence of the mixed-strategy PBE. Intuitively, when the lawyers' financial constraint are strong enough (i.e., when  $\tilde{A}$  is high enough and hence,  $\int_{\tilde{A}}^{\tau} (A + C_D)g(A)dA$  is low enough), then the mixed-strategy PBE emerges.

Condition (5) implies that the second derivative of the defendant's expected loss function  $l_D$ ,  $g(A) - C_D \frac{\partial g(A)}{\partial A}$ , is greater than zero, i.e.,  $l_D$  is strictly convex in the interval  $[\tilde{A}, \bar{A}]$ . Then, this condition ensures the uniqueness of the equilibrium positive settlement offer S. Condition (6) is a necessary condition for the existence of the mixed-strategy PBE with an equilibrium positive offer in the interval  $(\tilde{A}, \bar{A})$ , i.e., an interior positive offer.<sup>20</sup>

**Proposition 1.** Assume conditions (3)-(6) hold. The following strategy profile, together with the defendant's beliefs, characterize the mixed-strategy perfect Bayesian equilibrium.

(1) The defendant chooses a probability of accident  $\lambda_D = \arg \min \{K(\lambda) + \lambda \int_{\tilde{A}}^{\tilde{A}} (A + C_D)g(A)dA\}$ . If a lawsuit is filed, the defendant mixes between proposing a zero offer with probability  $\beta = 1 - \frac{f}{\gamma S}$ ; and, proposing an offer  $S \in (\tilde{A}, \bar{A})$ , implicitly defined by equations (1) and (2), with the complementary probability.

(2) A high-damage case is always filed by the plaintiff's lawyer; an average low-damage case is filed with probability  $\frac{\nu}{G(\tilde{A})}$ .

(3) A high-damage plaintiff always rejects a zero offer and accepts an offer S > 0 if and only if  $A \leq S$ ; a low-damage plaintiff always accepts a non-negative offer.

(4) The defendant's equilibrium beliefs are as follows. When the defendant observes a lawsuit, he believes that  $P(0 < A < \tilde{A}) = \frac{\nu}{\nu+1-G(\tilde{A})}$ ,  $P(\tilde{A} \le A \le y) = \frac{G(y)-G(\tilde{A})}{\nu+1-G(\tilde{A})}$  for any  $y \in [\tilde{A}, \bar{A}]$ .<sup>21</sup>

# Equilibrium No-Access to Justice $\eta$ : Formal Characterization

We now formally characterize the No-Access to Justice component  $\eta$  under the mixed-strategy PBE. Given Definition 2 and the equilibrium outcomes:

$$\eta = \left[\int_0^{\tilde{A}} g(A)dA - \nu\right] + \nu\beta.$$
(7)

Intuitively,  $\eta$  represents the inability of true victims to get access to justice. It takes into account (1) the mass of low-damage potential plaintiffs who cannot file a lawsuit,  $\int_0^{\tilde{A}} g(A) dA - \nu$ ;<sup>22</sup> and, (2) the mass of low-damage plaintiffs who file a lawsuit but receive a zero offer, and hence, need to drop their cases,  $\nu\beta$ . It is worth noting that ours is the first formal characterization of the access to justice concept in the literature.

Table 1 summarizes the equilibrium outcomes and no-access to justice component. Our previous analysis suggests that the lawyer's constraint  $\tilde{A}$  affects equilibrium strategies and outcomes

<sup>21</sup>See Lemma 7 in the Appendix for a detailed analysis of the defendant's equilibrium beliefs.

<sup>22</sup>Remember that 
$$\int_0^A g(A)dA - \nu = 1 - \zeta = 1 - \left\lfloor \int_{\tilde{A}}^A g(A)dA + \nu \right\rfloor$$
, where  $\zeta$  is the mass of filed cases.

<sup>&</sup>lt;sup>20</sup>Formal proofs of the mixed- and pure-strategy PBE with corner solutions (i.e., where the equilibrium positive settlement offer is equal to the highest possible plaintiff's type  $\bar{A}$ ) are presented in the Supplementary Material.

Table 1: Equilibrium Outcomes and No-Access to Justice Component (Mixed-Strategy PBE)

Probability of Zero Offer	$\beta = 1 - \frac{f}{\gamma S}$
Mass of Cases that Proceed to Trial	$\rho = \beta [1 - G(\tilde{A})] + (1 - \beta) [1 - G(S)]$
Probability of Filing	$\zeta =  u + \int_{ ilde{A}}^{ ilde{A}} g(A) dA$
Probability of Accident	$\lambda_D = \arg\min\{K(\lambda) + \lambda l_D\}$
Defendant's Expected Litigation Loss	$l_D = \int_{\tilde{A}}^{\bar{A}} (A + C_D) g(A) dA$
No-Access to Justice Component	$\eta = \left[\int_{0}^{\tilde{A}} g(A) dA - \nu ight] +  u eta$

*Notes*: Probability of filing conditional on accident occurrence; other outcomes conditional on accident occurrence and filing.

under the mixed-strategy PBE. Importantly, the mixed-strategy PBE and the characterization of the  $\eta$  component are aligned with the actual state of affairs in civil litigation. In real-word settings, lawyers confront financial constraints. As a result, only some low-damage victims get legal representation and file a lawsuit. Financial constraints also preclude lawyers to bring cases associated with low-damage clients to trial. Then, when defendants refuse to make a settlement offer, then lawyers are forced to drop the lawsuit. As a result, some victims get deprived of their civil right to access to justice.

# 3.2 Pure-Strategy PBE

When the sign of condition (4) is reversed, i.e., when  $G(\tau)\tau - \int_{\tilde{A}}^{\tau} (A + C_D)g(A)dA < 0$  for some  $\tau \in [\tilde{A}, \bar{A}]$ , then the pure-strategy PBE emerges. Intuitively, in a state of the world characterized by mild lawyers' financial constraints, then low-damage plaintiffs are relatively rare and high-damage plaintiffs are relatively common. The defendant will reduce his expected litigation loss by always making a positive offer because high-damage plaintiffs will always reject a zero offer and proceed to costly trial. Hence, when the level of lawyers' financial constraints is mild, then the defendant will have a strong incentive to always make a positive offer.

In the pure-strategy equilibrium, lawyers with all types of cases cases file lawsuits. Due to asymmetric information, some high-damage plaintiffs receive an insufficiently high offer (i.e., an offer lower than the actual harm from the accident) and proceed to costly trial, and all low-damage plaintiffs receive a generous offer (i.e., an offer greater than the actual harm from the accident) and settle out-of-court. Although attorneys with low-damage cases cannot proceed to trial due to their financial constraints, their expectations about a positive out-of-court settlement offer induce them to file a lawsuit. Given that all low-damage cases are filed, and all filed low-damage cases receive a positive offer, all victims get access to justice.

# Equilibrium Strategies

First, we analyze the potential composition of the set of equilibrium settlement offers and the potential composition of the equilibrium mass of filed cases. The proof of Lemma 8 shows that the set of equilibrium offers might only involve pure strategies with positive offers  $S \in [\tilde{A}, \bar{A}]$ . By condition (3), lawyers with high- and low-damage cases will get strictly positive expected payoffs. Then, all cases will be filed. Hence, the equilibrium mass of low-damage cases that are filed will be equal to the mass of low-damage cases,  $\nu = G(\tilde{A})$ , and the equilibrium mass of filed cases will be equal to the mass of potential plaintiffs,  $\zeta = 1$ .

We now demonstrate that the equilibrium settlement offer S exists and is unique. The defendant's expected litigation loss  $l_D(S) = G(S)S + \int_S^{\bar{A}} (A + C_D)g(A)dA$ . The first term, G(S)S, represents the loss associated with the acceptance of the out-of-court settlement offer S by lowand high-damage plaintiffs. The second term,  $\int_S^{\bar{A}} (A + C_D)g(A)dA$ , represents the loss associated with high-damage cases that proceed to trial.<sup>23</sup> Minimization of  $l_D(S)$  yields the first-order condition:

$$G(S) = C_D g(S). \tag{8}$$

When  $g(S) - C_D \frac{\partial g(S)}{\partial S} > 0$  for all  $S \in [\tilde{A}, \bar{A}]$ , the second-order optimality is satisfied. Then, the function is strictly convex and has a unique minimum. When  $1 - C_D g(\bar{A}) > 0$ , the optimal offer S is interior (i.e.,  $S \in (\tilde{A}, \bar{A})$ ).

Finally, we assess the optimal decision of the defendant at the care-taking stage (see Lemma 9 in the Appendix). By applying the logic used in case of the mixed-strategy PBE, we show that the defendant's equilibrium probability of an accident  $\lambda_D = \arg \min K(\lambda) + \lambda l_D$  exists, is unique, and is decreasing in  $l_D$ .

Table 2 summarizes the equilibrium outcomes and no-access to justice component under the pure-strategy PBE. Proposition 2 characterizes the pure-strategy PBE under conditions (3), (4'), (5) and (6'). Conditions (4') and (6') are as follows.

$$\min_{\tau \in [\tilde{A}, \bar{A}]} \left\{ G(\tau)\tau - \int_{\tilde{A}}^{\tau} (A + C_D)g(A)dA \right\} < 0.$$

$$(4')$$

 $<sup>^{23}</sup>$ The net award at trial also determines the likelihoods of out-court-settlement and trial Bebchuk (1984). Note, however, that neither filing nor care-taking precautions are endogenous in Bebchuk (1984).

Table 2: Equilibrium Outcomes and No-Access to Justice Component (Pure-Strategy PBE)

Probability of a Zero Offer	$\beta = 0$
Mass of Cases that Proceed to Trial	$\rho = [1 - G(S)]$
Probability of Filing	$\zeta = 1$
Probability of Accident	$\lambda_D = \arg\min\{K(\lambda) + \lambda l_D\}$
Defendant's Expected Litigation Loss	$l_D = SG(S) + \int_S^{\bar{A}} (A + C_D)g(A)dA$
No-Access to Justice Component	$\eta = 0$

*Notes*: Probability of filing conditional on accident occurrence; other outcomes conditional on accident occurrence and filing.

$$1 - C_D g(A) > 0.$$
 (6')

Condition (4') implies that for some positive offer  $\tau \in [\tilde{A}, \bar{A}]$ , the defendant's expected loss from making this offer,  $G(\tau)\tau + \int_{\tau}^{\bar{A}} (A+C_D)g(A)dA$ , is strictly lower than his expected loss from making a zero offer,  $\int_{\bar{A}}^{\bar{A}} (A+C_D)g(A)dA$ . This condition ensures that a zero offer is not made in equilibrium. Then, this is a necessary condition for the existence of the pure-strategy PBE. Condition (6') is a necessary condition for the existence of the pure-strategy PBE with an equilibrium positive offer in the interval  $(\tilde{A}, \bar{A})$ , i.e., an interior offer, i.e., an offer  $S \in (\tilde{A}, \bar{A})$ .

**Proposition 2.** Assume that conditions (3), (4'), (5) and (6') hold. The following strategy profile, together with the defendant's beliefs, characterize the pure-strategy perfect Bayesian equilibrium.

(1) The defendant chooses a probability of accident  $\lambda_D = \arg \min \{K(\lambda) + \lambda l_D(S)\}$ , where  $l_D(S) = G(S)S + \int_S^{\bar{A}} (A + C_D)g(A)dA$ . If a lawsuit is filed, the defendant always proposes an offer  $S \in (\tilde{A}, \bar{A})$ , implicitly defined by  $G(S) = C_D g(S)$ .

(2) All cases are filed by the plaintiff's lawyer.

(3) A high-damage plaintiff rejects an offer  $S \in (\tilde{A}, \bar{A})$  if and only if  $A \ge S$ ; a low-damage plaintiff always accepts  $S \in (\tilde{A}, \bar{A})$ .

(4) The defendant's equilibrium beliefs are as follows. When the defendant observes a lawsuit, he believes that  $P(0 < A \le y) = G(y)$ .<sup>24</sup>

# 3.3 An Illustration: Uniform Distribution of Damages

A simple example using a uniform distribution of damages illustrates the model's results. Suppose that the plaintiff's damage types A are uniformly distributed on the interval  $(0, \bar{A}]$ . Suppose also

 $<sup>^{24}</sup>$ See Lemma 10 in the Appendix for a detailed analysis of the defendant's equilibrium beliefs.

that the injurer's cost of care function is  $K(\lambda) = B[1 - \sqrt{1 - (\lambda - 1)^2}]$ , where B > 0 is the cost of care when  $\lambda = 0$ . In this example, conditions (4) and (4') become  $\tilde{A} > C_D(\sqrt{2} - 1)$  and  $\tilde{A} < C_D(\sqrt{2} - 1)$ , respectively.

Intuitively, in a state of the world where the the lawyers confront high constraints, i.e., where  $\tilde{A} > C_D(\sqrt{2} - 1)$ , the mixed-strategy PBE emerges. Using equation (1) and (2), it can be demonstrated that the defendant's equilibrium positive settlement offer is  $S = \sqrt{\tilde{A}^2 + 2\tilde{A}C_D}$ , which will be offered with probability  $1 - \beta = \frac{f}{\gamma S}$ , and the mass of low-damage cases that are filed  $\nu = \frac{\tilde{A} + C_D - S}{A}$ . Importantly, the no-access to justice component  $\eta = \frac{\tilde{A}}{A} - \nu + \nu\beta > 0$ . Hence, some low-damage plaintiffs will be deprived of their right of access to justice under the mixed-strategy PBE.

In contrast, in a state of the world where the the lawyers confront mild constraints, i.e., where  $\tilde{A} < C_D(\sqrt{2}-1)$ , the pure-strategy PBE emerges. Using equation (8), it can be shown that the equilibrium settlement offer is  $S = C_D$ , which is always offered by the defendant, i.e., it is offered with probability  $1 - \beta = 1$ . As a result, in addition to all high-damages cases, all low-damage cases are filed, i.e., the mass of low-damage cases that are filed is equal to the mass of low-damage plaintiffs  $\nu = \frac{\tilde{A}}{\tilde{A}} = G(\tilde{A})$ . Due to the absence of a zero offer in equilibrium and the filing of all low-damage cases, all victims will get access to justice.

Our previous analysis demonstrates that the lawyers' constraints represented by  $\tilde{A}$  do not affect the equilibrium strategies and outcomes under the pure-strategy PBE.<sup>25</sup> Hence,  $\tilde{A}$  does not affect social welfare. Importantly, all victims get access to justice under the pure-strategy PBE. Given that our main goal is to assess the effect of policies aimed at relaxing the lawyers' constraints on equilibrium outcomes, access to justice and social welfare, the focus of the next sections will be on the mixed-strategy PBE.

# 4 Effect of a Cost-Reducing Policy on Equilibrium Outcomes and Access to Justice

This section studies the effect of a cost-reducing policy on the equilibrium outcomes and access to justice under the mixed-strategy PBE. For instance, consider the effect of a policy aimed at reducing the lawyer's financial cost of loans r. Given the definition of  $\tilde{A}$ , a reduction in r lowers  $\tilde{A} \equiv \frac{C_P}{\gamma} + \frac{(C_P + f - x)r}{\gamma}$ . Importantly, a reduction in r also lowers  $FC \equiv \frac{(C_P + f - x)r}{\gamma}$ . Note also that

 $<sup>^{25}</sup>$ In fact, the only exogenous parameter that affects equilibrium strategies and outcomes under the pure-strategy PBE is the defendant's litigation cost  $C_D$ .

the lawyer's constraint  $\tilde{A}$  decreases (increases) if and only if the lawyer's financial constraint FC decreases (increases). Then, it is appropriate to perform comparative-statics analysis of costreducing policies aimed at lowering the lawyer's financial constraint through the analysis of the effects of changes in  $\tilde{A}$ .<sup>26</sup> Formal proofs are presented in the Appendix.

# 4.1 Effect on Equilibrium Outcomes

Proposition 3 summarizes the comparative statics results of a reduction in A.

**Proposition 3.** A reduction in  $\tilde{A}$ : (1) increases the expected litigation loss of the defendant  $l_D$ ; (2) reduces the probability of accident  $\lambda_D$ ; (3) reduces the positive out-of-court settlement offer S; (4) increases the probability of a positive offer  $(1 - \beta)$ ; (5) increases the mass of filed cases  $\zeta$ ; and (6) increases the probability of trial  $\rho$  if  $C_D < \tilde{A}$ .<sup>27</sup>

Intuitively, a relaxation of the lawyers' financial constraints increases the mass of filed cases  $\zeta$  by inducing more cases with low A to be filed. Due to the asymmetry of information between the plaintiff and the defendant, the higher likelihood of confronting plaintiffs with low A induces the defendant to lower the positive settlement offer S. Given the inverse relationship between the positive settlement offer and the probability of making that offer  $(1 - \beta) = \frac{f}{\gamma S}$ , the probability of making a positive offer increases. Although a positive offer is more frequently made, this offer is lower. As a result, the likelihood of trial  $\rho$  and hence, the expected litigation costs, increase. As a consequence of the negative impact of the higher likelihoods of filing and trial (and higher litigation costs), which are not offset by the positive effect of the lower positive settlement offer, the defendant's expected loss  $l_D$  increases and hence, his expenses on care also increase. This latter effect reduces the likelihood of accident  $\lambda_D$ .<sup>28</sup>

# 4.2 Effect on No-Access to Justice

The effect of a reduction in  $\tilde{A}$  on No-Access to Justice  $\eta$  are outlined in Proposition 4.

<sup>&</sup>lt;sup>26</sup>This analysis applies to the other four exogenous parameters encompassed in FC:  $C_P$ , f,  $\gamma$ , and x.

<sup>&</sup>lt;sup>27</sup>Intuitively, if the defendant's trial costs  $C_D$  are low enough, then the decrease in S due to a reduction in  $\tilde{A}$  will be strong enough to increase the probability of trial. This is a sufficient (but not necessary) condition.

<sup>&</sup>lt;sup>28</sup>Consider a binary-type version of our model. Assume that  $A_L < \tilde{A} < A_H$ , where  $A_L$  and  $A_H$  represent the low and high type, respectively. It is simple to show that the equilibrium outcomes do not depend on  $C_P$ , xand r. Then, comparative statics analysis with respect to the determinants of the lawyers' constraints cannot be performed using a binary-type model. These results apply to any discrete-type version of our model.

# **Proposition 4.** A reduction in $\tilde{A}$ reduces No-Access to Justice $\eta$ .

A relaxation of the lawyers' financial constraints increases access to justice by victims by affecting  $\eta$  in two ways. First, a reduction in  $\tilde{A}$  induces more cases with low A to be filed. Specifically, a lower  $\tilde{A}$  increases the mass of high-damage cases  $1-G(\tilde{A})$ : Some initially low-damage cases become high-damage cases. Remember that high-damage cases are always filed. Second, a reduction in  $\tilde{A}$  reduces the mass of low-damage cases that are dropped. Specifically, a lower  $\tilde{A}$  reduces the likelihood of a zero offer  $\beta$ .

# 5 Social Welfare and Deterrence

This section characterizes the social welfare loss function and presents an analysis of deterrence for the mixed-strategy PBE.

# 5.1 Social Welfare Loss Function

The social welfare loss function SWL includes the social welfare loss associated with the resources devoted to accident prevention  $K(\lambda)$ , and the unconditional social welfare loss from an accident  $\lambda l_W$ . The social welfare loss from an accident  $l_W$  includes the social welfare loss associated with the expected harm from an accident  $H = \int_0^{\bar{A}} Ag(A) dA$ , and the social welfare loss associated with the resources devoted to filing,  $\zeta f$ , and litigation,  $\rho(C_P + C_D)$ , conditional on accident occurrence. Second, it includes the social welfare loss associated with the expected infringement of citizens' right of access to justice,  $\theta\eta$ , where  $\theta \geq 0$  represents the society's concern regarding the citizens' right of access to justice.<sup>29</sup> Preserving the right of access to justice is an increasing concern for the U.S. and the international community. The U.S. Department of Justice established the Access to Justice Initiative (ATJ) in March 2010. Similarly, the United Nations considered for the first time including Access to Justice as part of its Sustainable Development Goals (Goal 16) in 2015.

**Definition 3.** The social welfare loss function SWL is defined as follows.

$$SWL(\lambda) = K(\lambda) + \lambda l_W, \tag{9}$$

where

$$l_W = H + \zeta f + \rho (C_P + C_D) + \theta \eta. \tag{10}$$

<sup>&</sup>lt;sup>29</sup>The higher  $\theta$ , the higher the social welfare loss associated with no-access to justice by victims. See Kornhauser (2015, 2003) for a discussion on economic analysis of law and welfare analysis of public policies. See also Chang (2000).

# 5.2 Analysis of Deterrence

This section presents an analysis of deterrence. We first formally define undeterrence and overdeterrence. We then identify the main components that affect deterrence. Finally, we present a numerical example to illustrate the deterrence components.

**Definition 4.** The defendant's deterrence level is defined as follows.

- (1) Underdeterrence:  $l_W l_D > 0$ .
- (2) Overdeterrence:  $l_W l_D < 0$ .

Let  $\lambda_W = \arg \min \{K(\lambda) + \lambda l_W\}$  represent the socially-optimal probability of accidents. By Lemma 6 in the Appendix,  $\lambda_W < \lambda_D$  if and only if  $l_W > l_D$ , and  $\lambda_W > \lambda_D$  if and only if  $l_W < l_D$ . Intuitively, underdeterrence involves private expenses on accident prevention below the socially optimal. As a result, the likelihood of accident is above the socially optimal. Conversely, overdeterrence involves private expenses on accident prevention above the socially optimal. Hence the likelihood of accident is below the socially optimal.

#### **Deterrence** Components

The defendant's expected litigation loss  $l_D$ , given by each side of equation (1), can be also expressed as:

$$l_D = (1 - \beta) \left[ \nu S + \int_{\tilde{A}}^{S} Sg(A) dA + \int_{S}^{\tilde{A}} (A + C_D)g(A) dA \right] + \beta \left[ \int_{\tilde{A}}^{\tilde{A}} (A + C_D)g(A) dA \right].$$

Then,  $l_W - l_D$  is given by:

$$= [H + \zeta f + \rho(C_P + C_D) + \theta\eta] - \left\{ (1 - \beta) \left[ \nu S + \int_{\tilde{A}}^{S} Sg(A)dA + \int_{S}^{\tilde{A}} (A + C_D)g(A)dA \right] + \beta \left[ \int_{\tilde{A}}^{\tilde{A}} (A + C_D)g(A)dA \right] \right\}.$$

 $l_W - l_D =$ 

Rearranging terms:

$$l_W - l_D =$$

$$= \left\{ \rho C_D - \left[ \beta \int_{\tilde{A}}^{S} g(A) dA + (1 - \beta) \int_{S}^{\tilde{A}} g(A) dA \right] C_D \right\} +$$

$$+ (\zeta f + \rho C_P + \theta \eta) +$$

$$+\left\{H-\left\{(1-\beta)\left[\nu S+\int_{\tilde{A}}^{S}Sg(A)dA+\int_{S}^{\tilde{A}}Ag(A)dA\right]+\beta\left[\int_{\tilde{A}}^{\tilde{A}}Ag(A)dA\right]\right\}\right\}.$$

Consider the first term (in curly brackets, first line). Note that  $\rho C_D$  represents the social loss associated with resources devoted to defendant's litigation costs;<sup>30</sup> and,  $\left[\beta \int_{\bar{A}}^{S} g(A) dA + (1 - \beta) \int_{S}^{\bar{A}} g(A) dA \right] C_D$  represents the defendant's expected loss associated with litigation costs paid at trial when zero and positive offers are rejected by high-damage plaintiffs. The social and private losses associated with the defendant's litigation costs should be equal. Then, this first term is equal to zero. Hence,

$$l_W - l_D =$$

$$= (\zeta f + \rho C_P + \theta \eta) + \left\{ H - \left\{ (1 - \beta) \left[ \nu S + \int_{\tilde{A}}^{S} Sg(A) dA + \int_{S}^{\tilde{A}} Ag(A) dA \right] \right\} + \beta \left[ \int_{\tilde{A}}^{\tilde{A}} Ag(A) dA \right] \right\} \right\}.$$
(11)

The first term (in parentheses, first line) includes the social loss associated with resources devoted to filing and plaintiff's litigation costs, and the social loss associated with the infringement of victims' right of access to justice. The defendant does not consider these three items. Then, this term is greater than zero.

The second term (in curly brackets, second line) includes the social loss associated with the expected harm from an accident H experienced by all victims (harm from an accident related to low-damage cases that are filed, low-damage cases that are not filed, and high-damage cases that are always filed) minus the expected compensation paid by the defendant (in the form of an out-of-court settlement amount or an award at trial). The second term might be greater or lower than zero.

The second term is positive when the harm caused by an accident is undercompensated, i.e., when H is greater than the expected compensation paid by the defendant. This occurs when undercompensation (to low-damage plaintiffs who receive a zero offer and are forced to drop their cases and to low-damage victims who cannot file a lawsuit) more than offsets overcompensation (to high- and low-damage plaintiffs who receive a positive offer greater than their actual damages). Conversely, the second term is negative when the harm caused by an accident is overcompensated, i.e., when the expected compensation paid by the defendant is greater than H. This occurs when overcompensation (to low- and high-damage plaintiffs who receive a positive offer greater than H. This occurs when their actual damages) more than offsets undercompensation (to low-damage plaintiffs who receive a positive offer greater than their actual damages) more than offsets undercompensation (to low-damage plaintiffs who receive a zero offer and are forced to drop their cases and low-damage victims who cannot file a lawsuit).

<sup>&</sup>lt;sup>30</sup>Remember that the equilibrium mass of cases that proceed to trial is  $\rho = \beta \int_{\tilde{A}}^{S} g(A) dA + (1-\beta) \int_{S}^{\tilde{A}} g(A) dA$ .

It is worth noting that infringement of some victims' right of access to justice will occur even in case of overcompensation. This result is aligned with real-world environments.

Hence, underdeterrence  $(l_W - l_D > 0)$  or overdeterrence  $(l_W - l_D < 0)$  might occur in our environment.<sup>31</sup> Proposition 5 establishes a sufficient condition for underdeterrence.

# **Proposition 5.** There exists $\bar{f} > 0$ such that, for any $f < \bar{f}$ , $l_W - l_D > 0$ .

The intuition is as follows. Remember that the probability that the defendant makes a positive offer is  $(1 - \beta) = \frac{f}{\gamma S}$ . Overcompensation to low-damage plaintiffs and to high-damage plaintiffs with  $A \in [\tilde{A}, S)$  might occur only when the defendant makes a positive out-of-court settlement offer.<sup>32</sup> Specifically, overcompensation to low-damage plaintiffs occurs with probability  $(1-\beta)\frac{\nu}{G(\tilde{A})}$ , and overcompensation to high-damage plaintiffs with  $A \in [\tilde{A}, S)$  occurs with probability  $(1 - \beta)$ . Then, low filing costs f imply a low probability of overcompensation, and hence, a low probability of overdeterrence. As we show in the Appendix, when f is low enough  $(f < \bar{f})$ , then the social loss from an accident  $l_W$  is greater than the defendant's private expected loss from litigation  $l_D$ . Hence, the defendant is underdeterred.

It is worth noting that, even if the defendant is overdeterred, when the concern of society about access to justice is strong enough (i.e., when  $\theta$  is large enough), the overall welfare effect of a cost-reducing policy will be always positive. (See Section 6.3 for details.)

#### A Numerical Example of Deterrence

We present a numerical example to illustrate the deterrence components. We use the uniformdistribution model presented in Section 3.3. We adopt the following set of exogenous parameters:  $\{C_P, C_D, \gamma, f, x, \theta, \overline{A}, B, r\} = \{200, 500, .33, 70, 100, 750, 1200, 1000, .18\}$ . The model conditions hold under the chosen parameter set. The MATLAB software is used to construct the numerical example.

The equilibrium outcomes under this numerical example are as follows. First, the social loss associated with resources devoted to filing and plaintiff's litigation costs and the social loss associated with the infringement of victims' right of access to justice are equal to 530 (first term of

<sup>&</sup>lt;sup>31</sup>When f is sufficiently low, which is likely to occur in real-world settings, then  $\beta = 1 - \frac{f}{\gamma S}$  is sufficiently close to unity. As a result,  $\{(1-\beta)[\nu S + \int_{\tilde{A}}^{S} Sg(A)dA + \int_{S}^{\tilde{A}} Ag(A)dA]\} + \beta [\int_{\tilde{A}}^{\tilde{A}} Ag(A)dA]\}$  is sufficiently close to  $[\int_{\tilde{A}}^{\tilde{A}} Ag(A)dA]$ . Given that  $H = \int_{0}^{\tilde{A}} Ag(A)dA$ , then the second term of equation (11) (in curly brackets, second line) will approach  $\int_{0}^{\tilde{A}} Ag(A)dA > 0$ . Hence, we might expect that  $l_W - l_D > 0$ . In other words, we might expect that underdeterrence will occur.

<sup>&</sup>lt;sup>32</sup>Remember that high-damage plaintiffs with  $A \in (S, \overline{A}]$  always reject an offer S and proceed to trial.

equation (11)). Note that these costs are not considered by the defendant. Second, the expected harm from an accident is 600 and the expected compensation paid by the defendant is only 428. In other words, the harm caused by an accident is undercompensated. The undercompensated harm is equal to 600 - 428 = 172 (second term of equation (11)). Then,  $l_W - l_D = 530 + 172 = 702$ . Hence, in this numerical example, the defendant is underdeterred.

# 6 Effect of a Cost-Reducing Policy on Social Welfare

We analyze the social welfare effect of a cost-reducing policy for the mixed-strategy PBE. Formal proofs are presented in the Appendix.

We show that a cost-reducing policy affects social welfare in several ways. First, a costreducing policy positively affects social welfare by increasing access to justice, and negatively affects social welfare by increasing the social costs associated with higher probabilities of filing and trial. In addition, a cost-reducing policy affects social welfare by increasing the defendant's care-taking incentives, and hence, by reducing the probability of accidents. This effect is positive if the defendant is initially underdeterred and hence the policy contributes to the alignment of the private and social probabilities of accident. We provide a necessary and sufficient condition for a positive overall welfare effect of a cost-reducing policy: When when society's concern of preserving access to justice for true victims is strong enough, then a cost-reducing policy is always welfare improving.

We start our analysis by decomposing the overall welfare effect into two components, indirect and direct effects. The indirect effect captures the impact of a cost-reducing policy on the potential injurer's care-taking incentives and hence, on the privately optimal probability of accidents  $\lambda_D$ . The direct effect captures the impact of a cost-reducing policy on the social loss from an accident  $l_W$ . Then, the indirect and direct effects are computed for a given  $l_W$  and a given  $\lambda_D$ , respectively.

Let  $SWL(\lambda_D) = K(\lambda_D) + \lambda_D l_W$  be the social welfare loss function evaluated at the privatelyoptimal probability of an accident  $\lambda_D$ . The overall welfare effect (OE) of a reduction in  $\tilde{A}$  is given by:

$$\frac{dSWL(\lambda_D)}{d\tilde{A}} = \frac{\partial SWL(\lambda_D)}{\partial \lambda_D} \frac{\partial \lambda_D}{\partial \tilde{A}} + \frac{\partial SWL(\lambda_D)}{\partial \tilde{A}}$$

The first term in the right-hand side of the equation represents the indirect effect (IE) while the second term represents the direct effect (DE).

# 6.1 Direct Effect (DE)

The direct effect can be computed by explicit differentiation:

$$\frac{\partial SWL(\lambda_D)}{\partial \tilde{A}} = \lambda_D f \frac{\partial \zeta}{\partial \tilde{A}} + \lambda_D (C_P + C_D) \frac{\partial \rho}{\partial \tilde{A}} + \lambda_D \theta \frac{\partial \eta}{\partial \tilde{A}}.$$
(12)

The direct effect includes two negative welfare effects: (1) The effect of larger filing costs due to higher probability of filing (first term in f;  $\frac{\partial \zeta}{\partial \tilde{A}} < 0$ , by Proposition 3); and, (2) the effect of larger litigation costs due to higher probability of trial (second term in  $C_P + C_D$ ;  $\frac{\partial \rho}{\partial \tilde{A}} < 0$ , by Proposition 3). It also includes a positive welfare effect: The effect on the No-Access to Justice component  $\eta$ . By Proposition 4,  $\frac{\partial \eta}{\partial \tilde{A}} > 0$ , i.e., a reduction in  $\tilde{A}$  lowers No-Access to Justice  $\eta$ . This positive effect offsets the negative effects if, for instance, the concern of society about preserving the citizens' right of access to justice is strong enough (i.e., if the value of  $\theta$  is large enough). In this case, the direct effect of a cost-reducing policy will be welfare improving.<sup>33</sup>

# 6.2 Indirect Effect (IE)

To compute the indirect effect  $\frac{\partial SWL(\lambda_D)}{\partial \lambda_D} \frac{\partial \lambda_D}{\partial \bar{A}}$ , first we take into account that at the point of the defendant's optimum,  $\lambda_D$ , the first-order optimality condition implies:

$$\frac{\partial K(\lambda_D)}{\partial \lambda_D} = -l_D$$

Differentiating the first-order optimality condition with respect to  $\hat{A}$  yields:

$$\frac{\partial \lambda_D}{\partial \tilde{A}} = -\frac{\frac{\partial l_D}{\partial \tilde{A}}}{\frac{\partial^2 K(\lambda_D)}{\partial \lambda_D^2}} = \frac{(\tilde{A} + C_D)g(\tilde{A})}{\frac{\partial^2 K(\lambda_D)}{\partial \lambda_D^2}}.^{34}$$

Second, we consider that:

$$\frac{\partial SWL(\lambda_D)}{\partial \lambda_D} = \left[\frac{\partial K(\lambda_D)}{\partial \lambda_D} + l_W\right] = -l_D + l_W$$

Then, the indirect effect can be written as:

$$\frac{\partial SWL(\lambda_D)}{\partial \lambda_D} \frac{\partial \lambda_D}{\partial \tilde{A}} = (l_W - l_D) \frac{\partial \lambda_D}{\partial \tilde{A}}$$

where  $(l_W - l_D)$  is given by equation (11). Hence, then the indirect effect can be written as:

$$\frac{\partial SWL(\lambda_D)}{\partial \lambda_D} \frac{\partial \lambda_D}{\partial \tilde{A}} =$$

<sup>34</sup>By Proposition 3,  $\frac{\partial \lambda_D}{\partial \tilde{A}} > 0$ .

<sup>&</sup>lt;sup>33</sup>When  $\theta = 0$ , the direct effect will be always welfare reducing.

$$= \left\{ \left(\zeta f + \rho C_P + \theta \eta\right) + \left\{ H - \left\{ (1 - \beta) \left[ \nu S + \int_{\tilde{A}}^{S} Sg(A) dA + \int_{S}^{\tilde{A}} Ag(A) dA \right] \right\} + \beta \left[ \int_{\tilde{A}}^{\tilde{A}} Ag(A) dA \right] \right\} \right\} \frac{\partial \lambda_D}{\partial \tilde{A}}.$$
 (13)

The indirect effect operates through a change in the defendant's care-taking incentives, i.e., through a change in  $l_D$  which affects the probability of accidents  $\lambda_D \left(\frac{\partial \lambda_D}{\partial \tilde{A}} \text{ term}\right)$ . Note, however, that it is also influenced by the access to justice component ( $\theta\eta$  term).

Given that  $\frac{\partial \lambda_D}{\partial A} > 0$  (by Proposition 3), the sign of the indirect effect depends on the sign of the term  $(l_W - l_D)$  (term multiplied by  $\frac{\partial \lambda_D}{\partial A}$ ). That is, the sign of the indirect effect depends on the relationship between the social loss from an accident and the defendant's expected litigation loss. If the defendant is initially underdeterred  $(l_D < l_W \text{ or } \lambda_D > \lambda_W)$ , then a reduction in  $\lambda_D$ will positively affect social welfare.<sup>35</sup> In other words, the indirect effect of a cost-reducing policy will be welfare enhancing by aligning the private and social probabilities of accidents. In contrast, if the defendant is initially overderdeterred  $(l_W < l_D \text{ or } \lambda_D < \lambda_W)$ , then the indirect effect of a cost-reducing policy will be welfare reducing.

# 6.3 Overall Effect(OE)

The overall effect of a cost-reducing policy is:

$$\frac{dSWL(\lambda_D)}{d\tilde{A}} = \left[ (l_W - l_D) \frac{\partial \lambda_D}{\partial \tilde{A}} \right] + \left[ \lambda_D f \frac{\partial \zeta}{\partial \tilde{A}} + \lambda_D (C_P + C_D) \frac{\partial \rho}{\partial \tilde{A}} + \lambda_D \theta \frac{\partial \eta}{\partial \tilde{A}} \right], \tag{14}$$

where  $(l_W - l_D)$  is given by equation (11). The first and second terms in brackets represent the indirect and direct effects of a cost-reducing policy, respectively.

It is worth noting that our framework generalizes seminal models of civil litigation (Bebchuk, 1984, 1988; Katz, 1990) by endogenizing the decisions associated with every step of a legal dispute. Then, our study provides detailed information of the effect of a cost-reducing policy on social welfare. As we discussed in the previous two sections, a cost-reducing policy affects social welfare through many individual forces, that although observed in real-world settings, produce in some cases opposite effects on social welfare. As result, the overall effect of a cost-reducing policy is generally ambiguous in our environment.

Proposition 6 provides a necessary and sufficient condition on  $\theta$  for a positive overall effect of a cost-reducing policy. Intuitively, this condition describes the state of the world where cost-reducing policies will be always welfare improving.<sup>36</sup>

<sup>&</sup>lt;sup>35</sup>In this case, a reduction in  $\tilde{A}$  will reduce the social welfare loss, and hence, will increase social welfare.

<sup>&</sup>lt;sup>36</sup>When  $\theta = 0$ , a cost-reducing policy will be welfare improving if the defendant is initially underdeterred and the positive indirect effect more than offsets the negative direct effect.

We start by isolating the terms in equation (14) that do not include  $\eta$  and the terms that include  $\eta$ . We first define  $\Omega$  as:

O -

$$\begin{split} \mathcal{M} &= \\ &\equiv \left\{ \left(\zeta f + \rho C_P\right) + \right. \\ &+ \left\{ H - \left\{ \left(1 - \beta\right) \left[ \nu S + \int_{\tilde{A}}^{S} Sg(A) dA + \int_{S}^{\tilde{A}} Ag(A) dA \right] \right\} + \beta \left[ \int_{\tilde{A}}^{\tilde{A}} Ag(A) dA \right] \right\} \right\} \right\} \frac{\partial \lambda_D}{\partial \tilde{A}} + \\ &+ \left[ \lambda_D f \frac{\partial \zeta}{\partial \tilde{A}} + \lambda_D (C_P + C_D) \frac{\partial \rho}{\partial \tilde{A}} \right], \end{split}$$

where the first two lines (terms multiplied by  $\frac{\partial \lambda_D}{\partial \tilde{A}}$ ) include the terms of the indirect effect that do not depend on  $\eta$ . The last line (in brackets) includes the terms of the direct effect that do not depend on  $\eta$ . Second, we define  $\Lambda$  as:  $\Lambda \equiv \eta \frac{\partial \lambda_D}{\partial \tilde{A}} + \lambda_D \frac{\partial \eta}{\partial \tilde{A}}$ . Then,

$$\Lambda \theta = \eta \frac{\partial \lambda_D}{\partial \tilde{A}} \theta + \lambda_D \frac{\partial \eta}{\partial \tilde{A}} \theta,$$

where the first component corresponds to the term of the indirect effect that depends on  $\eta$ , and the second component corresponds to the term of the direct effect that depends on  $\eta$ . Hence,

$$\frac{dSWL(\lambda_D)}{d\tilde{A}} = \Omega + \Lambda\theta.$$

Third, we define  $\underline{\theta} \equiv \frac{-\Omega}{\Lambda}$ .

## **Proposition 6.** A cost-reducing policy is welfare improving if and only $\theta > \underline{\theta}$ .

Intuitively, in a state of the world where society's concern about access to justice for victims is high enough  $(\theta > \underline{\theta})$ , the overall effect of a cost-reducing policy will be always welfare improving. Hence, cost-reducing policies might be recommended.

Finally note that, in this environment, the overall effect of a cost-reducing policy will be positive even if the defendant is initially overdeterred (i.e., even if IE < 0). The reason is that, when society's concern about preserving access to justice for victims is strong enough, the positive direct effect, driven by access to justice by more victims, will more than offset the negative effect associated with an increase in overdeterrence.

# 6.4 Effect of a Change in r on Social Welfare: A Numerical Example

We present a numerical example to illustrate the effect of a change in r (a specific cost-reducing policy) on social welfare.<sup>37</sup> We use the uniform-distribution model presented in Section 3.3 and

<sup>&</sup>lt;sup>37</sup>The MATLAB software is used to construct this numerical example.

	$\Delta r = .2818$	$\Delta r = .1808$
Direct Effect (DE) - Main Components	DE > 0	DE > 0
$\Delta \zeta$ (Probability of Filing)	+10.2%	+9.4%
$\Delta^{\underline{\rho}}_{\zeta}$ (Probability of Trial)	+2.5%	+1.9%
$\Delta \eta$ (No-Access to Justice)	-7.3%	-7.9%
$\bullet\Delta\nu$ (Mass of Low-Damage Cases that Are Filed)	+4.7%	+5.0%
• $\Delta\beta$ (Probability of Zero Offer)	-1.2%	-1.3%
Indirect Effect (IE) - Main Components	IE > 0	IE > 0
$\Delta l_D$ (Defendant's Expected Litigation Loss)	+9.5%	+8.3%
$\Delta \lambda_D$ (Probability of Accidents)	-6.6%	-6.3%
Overall Effect (OE)	OE > 0	OE > 0
$\Delta SWL$ (Social Welfare Loss)	-3.2%	-2.8%

Table 3: Effect of a Change in r on Social Welfare when  $\theta > \underline{\theta}$ 

consider a state of the world characterized by  $\theta > \underline{\theta}$ . The parameter set adopted is as follows: { $C_P, C_D, \gamma, f, x, \theta, \overline{A}, B$ } = {200, 500, .33, 70, 100, 750, 1200, 1000}. Three values of r are studied:  $r \in \{.08, .18., .28\}$ .<sup>38</sup>

Table 3 summarizes our findings. In this example, the direct and indirect welfare effects of a reduction in r are positive. The effect of changes in r on the main components associated with the direct and indirect welfare effects are reported. Our findings are consistent across changes in r. Consider, for instance, a reduction in r from 18% to 8%. This cost-reducing policy lowers social welfare loss by 3%. The result might be explained by the positive indirect effect of the policy that operates through the increase in the care-taking incentives for an initially underdeterred potential injurer (i.e., an 8% increase in  $l_D$ ; as result, a 6% reduction in the probability of accident  $\lambda_D$ ). The reduction in the social welfare loss can be also explained by the positive direct effect of the cost-reducing policy that operates through an 8% reduction in the mass of low-damage victims that do not get access to justice. The improvement in access to justice is driven by a 5% increase in the mass of low-damage victims who file lawsuits. It is also driven by the reduction in the probability of a zero offer by 1%, i.e., by the reduction in the likelihood that low-damage victims who file lawsuits will be forced to drop their cases.

Our results provide important policy implications. Specifically, our findings suggest that in a state of the world characterized by strong society's concern about access to justice by victims,

<sup>&</sup>lt;sup>38</sup>The model conditions and the condition  $\theta > \underline{\theta}$  hold under the chosen exogenous parameters. In this numerical example, SWL is a monotonically increasing function of r. The value of r that minimizes SWL is r = .08. The values for  $\underline{\theta} \in \{192, 158, 125\}$ , for  $r \in \{.08, .18, .28\}$ , respectively.

the negative direct effect of cost-reducing policies explained by the higher probabilities of filing and trial is more than offset by the positive direct effect of access to justice by more victims and the positive indirect effect of higher care-taking incentives for potential injurers initially underdeterred. As a result, the overall effect of cost-reducing policies is welfare increasing. Importantly, our analysis underscores the alignment of cost-reducing policies with society's goal of preserving citizens' right of access to justice.

# 7 Summary and Conclusions

This article presents a comprehensive economic analysis of civil litigation in the presence of financially-constrained lawyers. The decisions at all the stages associated with a legal dispute are endogenized. Our tractable framework allows for asymmetric information, financially-constrained lawyers, third-party lawyer lending institutions, and a continuum of plaintiff's types. We completely characterize the equilibria of the model. Our work provides important methodological contributions to the economic analysis of law by generalizing seminal economic models of civil litigation, providing the first formal definition of access to justice, and offering a complete welfare analysis of relevant public policies.

Our paper demonstrates that the state of the world regarding the financial constraints faced by lawyers determines whether the mixed- or the pure-strategy equilibrium emerges. In particular, the mixed-strategy equilibrium occurs under strong financial constraints. Importantly, we show that the financial constraints faced by lawyers permeate every decision made by the parties involved in a legal dispute. Accidents and pre-trial bargaining disagreement are observed in equilibrium, and access to justice is denied to some victims under the mixed-strategy equilibrium.

Our work provides significant lessons for policy makers. First, our welfare analysis of policies aimed at alleviating lawyers' financial constraints demonstrates that, in a state of the world characterized by high society's concerns about preserving the citizens' right of access to justice, these policies are always welfare improving. Moreover, our findings indicate that even if the potential injurers are initially overdeterred, these policies will enhance social welfare. Hence, cost-reducing policies might be recommended. Second, our study suggests that a comprehensive assessment of public policies associated with legal disputes should consider the effects of these policies on care-taking incentives for potential injurers, in addition to their effects on filing and litigation. Third, our analysis underscores the importance of incorporating the concept of access to justice to the study of the social welfare effect of relevant public policies. In future work, we plan to extend our model to study the optimal design of contracts between the plaintiff's lawyer and the third-party lender. The new framework will include the third-party lender as a fourth player, and will allow for uncertainty about the outcome at trial in the form of court errors. Recourse and non-recourse loans, as well as other contract terms, will be evaluated. Filing and access to justice, pretrial bargaining, and care-taking incentives for potential injurers will be assessed in this environment. These, and other extensions, remain fruitful areas for future research.

# References

- Avraham, R. and Wickelgren, A.L. 2014. "Third-Party Litigation Funding: A Signaling Model." DePaul Law Review, pp. 233-264.
- Bebchuk, L.A. 1988. "Suing Solely to Extract a Settlement Offer." Journal of Legal Studies, pp. 437-450.
- Bebchuk, L.A. 1984. "Litigation and Settlement under Imperfect Information." RAND Journal of Economics, pp. 404-415.
- Bell, P.A. and O'Connell, J. 1997. Accidental Justice: The Dilemmas of Tort Law. Yale University Press.
- Carter, T. 2004. "Cash Up Front. New Funding Sources Ease Financial Strains on Plaintiffs Lawyers." ABA Journal, October.
- Chang, H. 2000. "A Liberal Theory of Social Welfare." Yale Law Journal, pp. 173-235.
- Choi, A. 2003. "Allocating Settlement Authority under a Contingent-Fee Arrangement." Journal of Legal Studies, pp. 586-610.
- Daughety, A.F. and Reinganum, J. F. 2014. "The Effects of Third-Party Funding of Plaintiffs on Settlement." American Economic Review, pp. 2552-2566.
- Daughety, A.F. and Reinganum, J. F. 2012. "Settlement." In Sanchirico, C.W. (ed.), The Encyclopedia of Law and Economics. E. Elgar Publishing.
- Deffains, B. and Desrieux, C. 2015. "To Litigate of Not to Litigate? The Impacts of Third-Party Financing on Litigation." International Review of Law and Economics, pp. 178–189.
- Demougin, D. and Maultzsch, F. 2014. "Third-Party Financing of Litigation: Legal Approaches and a Formal Model." CESifo Economic Studies, pp. 525–553.
- Emons, W. 2007. "Conditional versus Contingency Fees." Oxford Economic Papers, pp. 89-101.
- Engstrom, N.F. 2014. "Lawyer Lending: Costs and Consequences." DePaul Law Review, pp. 377-446
- Faughnan, B.S., Richmond, D.R., and Matula, M.L. 2011. Professional Responsibility in Litigation. Chicago: ABA Book Publishing.
- Frankel, A. 2006. "Helping Underfunded Plaintiffs Lawyers At a Price. A New Species of Lender Emerges to Help Smaller Plaintiffs Firms. But Is There A Catch?" The American Lawyer, February 13.
- Garber, S. 2010. "Alternative Litigation Financing in the United States: Issues, Knowns, and Unknowns." RAND Institute for Civil Justice Law, Finance, and Capital Markets Program, Occasional Paper.
- Glater, J. D. 2009. "Investing in an Portfolio of Lawsuits." *The New York Times.* Wednesday, June 3, 2009, p. B1 and p. 8.
- Katz, A. 1990. "The Effect of Frivolous Lawsuits on the Settlement of Litigation." International Review of Law and Economics, pp. 3-27.
- Kornhauser, L. 2015. "The Economic Analysis of Law." In Zalta, E.N. (ed.), Stanford Encyclopedia of Philosophy. http://plato.stanford.edu/archives/sum2015/entries/legal-econanalysis/.

- Kornhauser, L. 2003. "Preference, Well-Being, and Morality in Social Decisions." Journal of Legal Studies, pp. 303-329.
- Landeo, C.M. Forthcoming. "Law and Economics and Tort Litigation Institutions: Theory and Experiments." In Zeiler, K. and Teitelbaum, J. (eds.), *The Research Handbook on Behavioral Law and Economics*. E. Elgar Publishing.
- Landeo, C.M., Nikitin, M., and Babcock, L. 2007a. "Split-Awards and Disputes: An Experimental Study of a Strategic Model of Litigation." Journal of Economic Behavior and Organization, pp. 553-572.
- Landeo, C.M., Nikitin, M., and Baker, S. 2007b. "Deterrence, Lawsuits and Litigation Outcomes under Court Errors." Journal of Law, Economics and Organization, pp. 57-97.
- Landeo, C.M. and Nikitin, M. 2006. "Split-Award Tort Reform, Firm's Level of Care and Litigation Outcomes." Journal of Institutional and Theoretical Economics, pp. 571-600.
- Landsman, S., Abramowicz, M., Avraham, R., Wickelgren, A., Baker, L., DeStefano, M., Engstrom, N.F., Gilles, M., Sebok, A., Gold, A.S., Hensler, D. R., Hylton, K.N., Hyman, D.A., Silver, C., Ratner, M.A., Rubenstein, W.B., Wendel, W.B., and Yoon, A. 2014. "A Brave New World: The Changing Face of Litigation and Law Firm Finance." Nineteenth Annual Clifford Symposium on Tort Law and Social Policy." DePaul Law Review, pp. 195-718.
- McGovern, G., Rickman, N., Doherty, J.W., Kipperman, F., Morikawa, J., and Giglio, K. 2010. Third-Party Litigation Funding and Claim Transfer. Trends and Implications for the Civil Justice System. RAND Institute for Civil Justice Conference Proceedings.
- Miceli, T.J. 1994. "Do Contingency Fees Promote Excessive Litigation?" Journal of Legal Studies, pp. 211-224.
- Miller, J. 1994. "Some Agency Problems in Settlement." Journal of Legal Studies, pp. 189-215.
- Molot, J. T. 2010. "Litigation Finance: A Market Solution to a Procedural Problem." Georgetown Law Journal, pp. 65-115.
- Molot, J. T. 2009. "A Market in Litigation Risk." University of Chicago Law Review, pp. 367-439.
- Png, I.P.L. 1987. "Litigation, Liability, and the Incentives for Care." Journal of Public Economics, pp. 61-85.
- Png, I.P.L. 1983. "Strategic Behaviour in Suits, Settlement, and Trial." Bell Journal of Economics, pp. 539-550.
- Reinganum, J.F. and Wilde, L. 1986. "Settlement, Litigation, and the Allocation of Litigation Costs." RAND Journal of Economics, pp. 557-566.
- Rodak, M. 2006. "It's About Time: A Systems Thinking Analysis of the Litigation Finance Industry and its Effect on Settlement,." University of Pennsylvania Law Review, pp. 503-535.
- Rosenberg, D. and Shavell, S. 1985. "A Model in which Suits Are Brought for their Nuisance Value." International Review of Law and Economics, pp. 3-13.
- Shavell, S. 1982. "Suit, Settlement, and Trial: A Theoretical Analysis under Alternative Methods for the Allocation of Legal Costs." *Journal of Legal Studies*, pp. 55-81.

- Shepherd, J. 2014. "Uncovering the Silent Victims of the American Medical Liability System." Vanderbilt Law Review, pp. 151-195.
- Spier, K.E. and Dana, J. 1993. "Expertise and Contingency Fees: The Role of Asymmetric Information in Attorney Compensation." Journal of Law, Economics, and Organization, pp. 349–367.
- Steinitz, M. 2012. "The Litigation Finance Contract." William and Mary Law Review, pp. 455-517.
- The Economist. 2013. "Investing in Litigation. Second-Hand Suits." April 6th, pp. 72-73
- Trauner, M.T. 2009. "Personal Responsibility v. Corporate Liability: How Personal Injury Lawyers Screen Cases in an Era of Tort Reform." In Sandefur, R. (ed.), Sociology of Crime, Law and Deviance: Access to Justice. Bingley, UK: Emerald.

# Appendix: Mixed- and Pure-Strategy PBE

Define  $\hat{A} \equiv \frac{C_P + f + (C_P + f - x)r}{\gamma}$ .<sup>1</sup>

Claim 1. Let  $\check{A} \equiv \inf\{A_{filed}\}, \hat{A} > \check{A} \ge \check{A}$ , and that defendant makes a zero offer or an offer  $S \in (0, \tilde{A})$ . Then, n(A) = 1 for  $A > \check{A}$ .

**Proof.** By definition of  $\hat{A}$ , a case with  $A \ge \hat{A}$  is always filed. Remember that  $n(A) \in [0, 1]$  is the probability that a type A will be filed, for  $A \in [\tilde{A}, \hat{A}]$ . By assumption,  $\hat{A} > \check{A} \ge \tilde{A}$ . Then, there are two possible options: (1) Type  $\check{A}$  is filed, and (2) type  $\check{A}$  is not filed. Let b be the probability that a zero offer or an offer  $S_1 \in (0, \check{A})$  is made by the defendant; and (1 - b) be the probability that an offer  $S_2 \in [\check{A}, \bar{A}]$  is made by the defendant.<sup>2</sup> By assumption, the defendant makes a zero offer or an offer  $S \in (0, \tilde{A})$ ; then, b > 0. A zero offer or an offer  $S_1 \in (0, \check{A})$  will be always rejected by a plaintiff with type  $\check{A}$ . Consider option (1). Given that an attorney with a case  $\check{A}$  files a lawsuit by assumption, then it must be the case that his expected payoff is non-negative:  $b[\gamma\check{A} - C_P - (C_P + f - x)r] + (1 - b)\gamma S_2 - f \ge 0$ . Then, the expected payoff for an attorney with a case  $A > \check{A}$ :  $b[\gamma A - C_P - (C_P + f - x)r] + (1 - b)\gamma S_2 - f \ge 0$ . An attorney with a case  $A > \check{A}$  will always file a lawsuit. Hence, n(A) = 1. Now consider option (2). Given that  $\check{A} \equiv \inf\{A_{filed}\}$ , in any neighborhood of  $\check{A}$  can be infinitely small. Then, all  $A > \check{A}$  are filed. Then, all  $A > A_1$  are also filed. The neighborhood of  $\check{A}$  can be infinitely small.

# 1 Mixed-Strategy PBE – Proofs

Assume that conditions (3)-(6) hold.

**Claim 2.** Suppose that condition (4) holds. Suppose that the defendant mixes between a zero offer and a positive offer  $S \in [\tilde{A}, \bar{A}]$ . Then,  $G(\tilde{A}) > \nu$ .

**Proof.** By assumption, the defendant mixes between a zero offer and a positive offer  $S \in [\tilde{A}, \bar{A}]$ . Then, he must be indifferent between these two offers:  $\int_{\tilde{A}}^{\tilde{A}} (A + C_D)g(A)dA = \nu S + \int_{\tilde{A}}^{S} Sg(A)dA + \int_{S}^{\tilde{A}} (A + C_D)g(A)dA$ , where the left- and right-hand sides correspond to the defendant's expected litigation loss when he makes a zero offer and an offer  $S \in [\tilde{A}, \bar{A}]$ , respectively. By condition (4), for any  $S \in [\tilde{A}, \bar{A}]$ , including the positive offer that satisfies the indifference condition above:  $G(S)S > \int_{\tilde{A}}^{S} (A + C_D)g(A)dA$ . Then,  $G(\tilde{A})S + \int_{\tilde{A}}^{S} Sg(A)dA + \int_{S}^{\tilde{A}} (A + C_D)g(A)dA > \nu S + \int_{\tilde{A}}^{S} Sg(A)dA + \int_{S}^{\tilde{A}} (A + C_D)g(A)dA$ , for the positive offer that satisfies the indifference condition above. Hence,  $G(\tilde{A}) > \nu$ .

**Lemma 1.** The set of equilibrium offers must include a zero offer and at least one positive offer  $S \in [\tilde{A}, \bar{A}]$ .

**Proof.** First, it is simple to show that offers greater than  $\overline{A}$  are strictly dominated by an offer equal to  $\overline{A}$ . Second, we will demonstrate that offers  $S \in (0, \widetilde{A})$  are not in the set of equilibrium offers.<sup>3</sup> Suppose not. An offer  $S \in (0, \widetilde{A})$  is an equilibrium offer. By definition of  $\widehat{A}$ , a case with  $A \ge \widehat{A}$  is always filed. Remember also that  $n(A) \in [0, 1]$  is the probability that a type A will be filed, for  $A \in [\widetilde{A}, \widehat{A}]$ . Let the mass of low-damage cases that are filed be  $\nu \in [0, G(\widetilde{A})]$ . Let  $\epsilon > 0$  denote a small number. Consider the

 $<sup>{}^{1}\</sup>hat{A}$  refers to the damage threshold at which the lawyer's net payoff is zero when a high-damage case is solved at trial.

<sup>&</sup>lt;sup>2</sup>An offer  $S_2 \in [\check{A}, \bar{A}]$  must be considered to ensure that a lawyer with type  $A < \hat{A}$  will file.

 $<sup>{}^{3}</sup>S$  might be non-unique, i.e., multiple S might be made in equilibrium with positive probabilities.

case when  $\nu \in (0, G(\tilde{A})]$ . When the defendant makes an offer  $S \in (0, \tilde{A})$ , his expected litigation loss is  $\nu S + \int_{\tilde{A}}^{\tilde{A}} (A + C_D)n(A)g(A)dA + \int_{\tilde{A}}^{\tilde{A}} (A + C_D)g(A)dA$ . The defendant is better off by deviating to a zero offer, which generates an expected litigation loss equal to  $\int_{\tilde{A}}^{\tilde{A}} (A + C_D)n(A)g(A)dA + \int_{\tilde{A}}^{\tilde{A}} (A + C_D)g(A)dA$ . Contradiction follows. Consider now the case when  $\nu = 0$ . Let  $\tilde{A} \equiv \inf\{A_{filed}\}$ , where  $A_{filed}$  is the set of values of A that are filed and  $\hat{A} \geq \tilde{A} \geq \tilde{A}$ . Two options are possible: (1)  $\tilde{A} = \hat{A}$  and (2)  $\tilde{A} < \hat{A}^{.4}$  Under option (1), the defendant's expected litigation loss from offering  $S \in (0, \tilde{A})$  and offering  $\hat{A} + \epsilon$  are  $\int_{\tilde{A}}^{\hat{A}+\epsilon} (A + C_D)g(A)dA + \int_{\tilde{A}+\epsilon}^{\tilde{A}} (A + C_D)g(A)dA + \int_{\tilde{A}+\epsilon}^{\tilde{A}+\epsilon} (A + C_D)g(A)dA$ , respectively. The defendant will be better off by deviating and offering  $\hat{A} + \epsilon$ :  $\int_{\tilde{A}}^{\hat{A}+\epsilon} (A + C_D)g(A)dA$ , when  $C_D > \epsilon$ . Contradiction follows. Under option (2), the defendant's expected litigation loss from offering  $S \in (0, \tilde{A})$  and offering  $\tilde{A} + \epsilon$  will be  $\int_{\tilde{A}}^{\tilde{A}+\epsilon} (A + C_D)n(A)g(A)dA + \int_{\tilde{A}+\epsilon}^{\tilde{A}} (A + C_D)g(A)dA$ . His expected litigation loss from offering  $S \in (0, \tilde{A})$  will be  $\int_{\tilde{A}}^{\tilde{A}+\epsilon} (A + C_D)n(A)g(A)dA + \int_{\tilde{A}+\epsilon}^{\tilde{A}} (A + C_D)g(A)dA$ . His expected litigation loss from offering  $\tilde{A} + \epsilon$  will be  $\int_{\tilde{A}}^{\tilde{A}+\epsilon} (\tilde{A} + \epsilon)n(A)g(A)dA + \int_{\tilde{A}+\epsilon}^{\tilde{A}} (A + C_D)g(A)dA$ . By Claim 1, n(A) = 1, for  $A > \tilde{A}$ . The defendant will be better off by deviating and offering  $\tilde{A} + \epsilon$ :  $\int_{\tilde{A}}^{\tilde{A}+\epsilon} (\tilde{A} + \epsilon)g(A)dA < \int_{\tilde{A}}^{\tilde{A}+\epsilon} (A + C_D)g(A)dA$ , when  $C_D > \epsilon$ . Contradiction follows. Hence, an offering  $\tilde{A} + \epsilon$ :  $\int_{\tilde{A}}^{\tilde{A}+\epsilon} (\tilde{A} + \epsilon)g(A)dA < \int_{\tilde{A}}^{\tilde{A}+\epsilon} (A + C_D)g(A)dA$ , when  $C_D > \epsilon$ . Contradiction follows. Hence, an offer  $S \in (0, \tilde{A})$  cannot be in the set of equilibrium offers.

Third, we will demonstrate that a zero offer is in the set of equilibrium offers. Suppose not. There is not a zero offer in equilibrium. Then the defendant offers  $S \in [\tilde{A}, \bar{A}]$ .<sup>5</sup> By condition (3), all lowand high-damage cases will be filed. The defendant's expected litigation loss from offering S will be  $G(S)S + \int_{S}^{\bar{A}} (A + C_D)g(A)dA$ . By condition (4),  $G(S)S + \int_{S}^{\bar{A}} (A + C_D)g(A)dA > \int_{\bar{A}}^{\bar{A}} (A + C_D)g(A)dA$ , where the right-hand side of the inequality represents the defendant's expected litigation loss from making a zero offer. Then, the defendant's is better off by deviating to a zero offer. Contradiction follows. Hence, a zero offer must be in the set of equilibrium offers. As a result, the probability that the defendant makes a zero offer  $\beta > 0$ . Fourth, we will show that a zero offer cannot be the only equilibrium offer. Suppose not. The defendant always makes a zero offer. Then, only cases with types  $A \ge \hat{A}$  will be filed. These plaintiffs will always reject a zero offer. The defendant will be better off by deviating and offering  $\hat{A} + \epsilon$ :  $\int_{\hat{A}}^{\hat{A}+\epsilon} (\hat{A} + \epsilon)g(A)dA + \int_{\hat{A}+\epsilon}^{\bar{A}} (A + C_D)g(A)dA < \int_{\hat{A}}^{\hat{A}+\epsilon} (A + C_D)g(A)dA + \int_{\hat{A}+\epsilon}^{\bar{A}} (A + C_D)g(A)dA$ , when  $C_D > \epsilon$ . Contradiction follows. Hence, in addition to a zero offer, there must be at least one positive offer  $S \in [\tilde{A}, \bar{A}]$  in the set of equilibrium offers.  $\blacksquare$ 

**Lemma 2.** There must be at least some low-damage cases that are filed in equilibrium. All high-damage cases are filed in equilibrium.

**Proof.** First, we will demonstrate that there must be at least some low damage cases that are filed in equilibrium,  $\nu > 0$ . Suppose not. Low-damage cases are never filed. By definition of  $\hat{A}$ ,  $A \ge \hat{A}$  always files. By assumption,  $\hat{A} \ge \check{A} \ge \check{A}$ . Then, there are two possible options: (1)  $\check{A} = \hat{A}$ , and (2)  $\check{A} < \hat{A}$ . Consider option 1. The defendant will be better off by deviating from the mixed-strategy involving a zero offer and a positive offer  $S \in [\tilde{A}, \bar{A}]$  to a pure-strategy involving an offer  $\hat{A} + \epsilon$ :  $\int_{\hat{A}}^{\hat{A}+\epsilon} (\hat{A} + \epsilon)g(A)dA + \int_{\hat{A}+\epsilon}^{\bar{A}} (A + C_D)g(A)dA < \int_{\hat{A}}^{\hat{A}+\epsilon} (A + C_D)g(A)dA + \int_{\hat{A}+\epsilon}^{\bar{A}} (A + C_D)g(A)dA$ , when  $C_D > \epsilon$ . Consider option 2. The defendant's expected litigation loss from the mixed-strategy involving a zero offer and a positive offer  $S \in [\tilde{A}, \bar{A}]$  will be  $\int_{\tilde{A}}^{\tilde{A}+\epsilon} (A + C_D)n(A)g(A)dA + \int_{\tilde{A}+\epsilon}^{\bar{A}} (A + C_D)g(A)dA$ . His expected litigation loss from offering  $\check{A} + \epsilon$  will be  $\int_{\tilde{A}}^{\check{A}+\epsilon} (\check{A}+\epsilon)n(A)g(A)dA + \int_{\tilde{A}+\epsilon}^{\bar{A}} (A + C_D)g(A)dA$ . By

<sup>&</sup>lt;sup>4</sup>In words, at least for some  $A \in [\tilde{A}, \hat{A})$  are filed, n(A) > 0.

 $<sup>{}^{5}</sup>S$  might be non-unique, i.e. multiple S might be made in equilibrium with positive probabilities.

Claim 1 in Appendix A, n(A) = 1, for  $A > \check{A}$ . The defendant will be better off by deviating and offering  $\check{A} + \epsilon$ :  $\int_{\check{A}}^{\check{A}+\epsilon} (\check{A} + \epsilon)g(A)dA < \int_{\check{A}}^{\check{A}+\epsilon} (A + C_D)g(A)dA$ , when  $C_D > \epsilon$ . Then, a mixed-strategy with a zero offer and a positive offer  $S \in [\check{A}, \hat{A}]$  is not in the set of equilibrium offers. Contradiction follows. Hence, v > 0.

Second, we will show that all high-damage cases are filed. We just demonstrated that at least some low-damage cases are filed in equilibrium. Then, it must be the case that at least some attorneys with low-damage cases will get non-negative expected payoffs:  $\gamma[\beta(0) + (1 - \beta)S)] - f \ge 0$ , where S and  $\beta$  represent a positive offer and the probability of getting a zero offer, respectively.<sup>6</sup> We also showed that  $\beta > 0$  in equilibrium. Consider now the case of an attorney with  $A \ge \tilde{A}$ . The expected payoff for the attorney is  $\beta[\gamma A - C_P - (C_P + f - x)r] + (1 - \beta)\gamma S - f$ . This expression is non-negative for  $A = \tilde{A}$  and positive for  $A > \tilde{A}$ . As a consequence, all high-damage cases will be filed. Hence, the equilibrium mass of cases that are filed  $\zeta = \nu + \int_{\tilde{A}}^{\tilde{A}} g(A) dA$ , where  $\nu > 0$ .

**Lemma 3.** The system of equations (1)–(2) has a unique solution  $(S, \nu)$ , where  $S \in (\tilde{A}, \bar{A})$ .

**Proof.** Inserting equation (2) into (1) yields  $\int_{\tilde{A}}^{S} (A + C_D - S)g(A)dA - S[C_Dg(S) - G(S) + G(\tilde{A})] = 0.$ After simplification,  $\int_{\tilde{A}}^{S} (A + C_D - S)g(A)dA - S[C_Dg(S) - G(S) + G(\tilde{A})] = \int_{\tilde{A}}^{S} Ag(A)dA + C_D[G(S) - G(\tilde{A}) - Sg(S)].$  Denote  $\Phi(S) \equiv \int_{\tilde{A}}^{S} Ag(A)dA + C_D[G(S) - G(\tilde{A}) - Sg(S)].$  The proof proceeds in several steps: (1)  $\Phi(S)$  is continuous on the interval  $[\tilde{A}, \bar{A}]$ ; (2)  $\frac{\partial \Phi(S)}{\partial S} > 0$ ; (3)  $\Phi(\tilde{A}) < 0$ ; (4)  $\Phi(\bar{A}) > 0$ ; (5)  $\Phi(S) = 0$  has exactly one solution.

(1) Continuity of  $\Phi(S)$  follows from the assumptions about G(A) and g(A) functions. (2) Differentiation and further algebraic transformations yield  $\frac{\partial \Phi(S)}{\partial S} = S\left[g(S) - C_D \frac{\partial g(S)}{\partial S}\right] > 0$ . The last inequality follows from condition (5). (3)  $\Phi(\tilde{A}) = -\tilde{A}C_D g(\tilde{A}) < 0$ . (4) By condition (4),  $\Phi(\bar{A}) = \int_{\tilde{A}}^{\bar{A}} Ag(A) dA + C_D [1 - G(\tilde{A}) - \bar{A}g(\bar{A})] > 0$ . (5) We have showed that  $\Phi(S)$  is a strictly increasing and continuous function with  $\Phi(\tilde{A}) < 0$  and  $\Phi(\bar{A}) > 0$ . Hence, there exists a unique  $S \in (\tilde{A}, \bar{A})$  such that  $\Phi(S) = 0$ . By equation (2), existence and uniqueness of S implies existence and uniqueness of  $\nu$ .

**Lemma 4.** The equilibrium positive settlement offer  $S \in (\tilde{A}, \bar{A})$  and the equilibrium mass of low-damage cases that are filed  $\nu$ , which are implicitly defined by equations (1) and (2), exist and are unique.

**Proof.** We have established that a zero offer and at least one positive offer are made in equilibrium. This implies that the positive equilibrium offer S must minimize the expected litigation loss of the defendant, and the defendant must be indifferent between offering the optimal positive offer S and a zero offer. The indifference condition is given by equation (1). The first-order optimality condition simplifies to equation (2). Lemma 3 demonstrates that the system of equations (1)-(2) has a unique solution, S and  $\nu$ . By condition (5), the second-order optimality condition  $g(A) > C_D \frac{\partial g(A)}{\partial A}$  is satisfied for all  $A \in [\tilde{A}, \bar{A}]$ . By Claim 2,  $\nu < G(\tilde{A})$ .

**Lemma 5.** The equilibrium probability that the defendant makes a zero offer is  $\beta = 1 - \frac{f}{\gamma S}$ .

**Proof.** An attorney with an average low-damage client mixes between filing and not filing.<sup>7</sup> Then, he must be indifferent between these two strategies:  $f = \gamma[\beta(0) + (1 - \beta)S]$ . This indifference condition allows us to compute the equilibrium  $\beta$ :  $\beta = 1 - \frac{f}{\gamma S}$ .

<sup>&</sup>lt;sup>6</sup>The mixed-strategy equilibrium might involve a zero offer and multiple positive offers.

<sup>&</sup>lt;sup>7</sup>In principle, the probability of filing for a low-damage plaintiff may depend on the specific A. Then, the expression "average low-damage client" is used here.

**Lemma 6.** For any positive value of l, the function  $K(\lambda) + \lambda l$  has a unique interior minimum,  $\lambda^* \in (0, 1)$ , which is decreasing in l.

**Proof.** By the assumptions on  $K(\lambda)$ , the first derivative of  $K(\lambda) + \lambda l$  is continuous and strictly increasing, negative when  $\lambda$  approaches zero, and positive when  $\lambda$  approaches 1. Hence,  $K(\lambda) + \lambda l$  has a unique interior minimum,  $\lambda^*$ . Totally differentiating the first derivative of  $K(\lambda) + \lambda l$ ,  $\frac{\partial^2 K(\lambda)}{\partial \lambda^2} d\lambda + dl = 0$ . Hence,  $\frac{\partial \lambda}{\partial l} = -\frac{1}{\frac{\partial^2 K(\lambda)}{\partial \lambda^2}} < 0$ .

**Lemma 7.** In equilibrium, the defendant's posterior beliefs are as follow:  $P(0 < A < \tilde{A}|Filing) = \frac{\nu}{\nu+1-G(\tilde{A})}$ ,  $P(\tilde{A} \le A \le y|Filing) = \frac{G(y)-G(\tilde{A})}{\nu+1-G(\tilde{A})}$  for any  $y \in [\tilde{A}, \bar{A}]$ .

**Proof.** All high-damage cases and a subset of low-damage cases are filed. The mass of high-damage cases that are filed is equal to the total mass of high-damage cases,  $1 - G(\tilde{A})$ .  $\nu$  includes only the subset of low-damage cases that are filed. Therefore, the total mass of filed cases is  $1 - G(\tilde{A}) + \nu$ . The posterior (conditional on filing) probability of an average low-damage case is  $\frac{\nu}{1-G(\tilde{A})+\nu}$ , while the posterior (conditional on filing) probability of an average high-damage case is  $\frac{1-G(\tilde{A})}{1-G(\tilde{A})+\nu}$ . Given that all high-damage cases are filed and the distribution G(A) is known by the defendant, he computes the posterior probability for any range of high-damage cases as the product of conditional probability of a high-damage case) and the posterior probability of a high-damage case) and the posterior probability of a high-damage case. Hence, for any  $y \in (\tilde{A}, \bar{A}]$ ,  $P(\tilde{A} < A \le y | Filing) = \frac{G(y)-G(\tilde{A})}{1-G(\tilde{A})+\nu} = \frac{G(y)-G(\tilde{A})}{1-G(\tilde{A})+\nu}$ .

**Proposition 3.** A reduction in  $\tilde{A}$ : (1) increases the expected litigation loss of the defendant  $l_D$ ; (2) reduces the probability of an accident  $\lambda_D$ ; (3) reduces the positive out-of-court settlement offer S; (4) reduces the probability of a zero offer  $\beta$ ; (5) increases the mass of filed cases  $\zeta$ ; and (6) increases the probability of trial  $\rho$  if  $C_D < \tilde{A}$ .

#### Proof.

(1) The defendant's expected litigation loss is  $l_D = \int_{\tilde{A}}^{\tilde{A}} (A+C_D)g(A)dA$ . Then,  $\frac{\partial l_D}{\partial \tilde{A}} = -(\tilde{A}+C_D)g(\tilde{A}) < 0$ . (2) By Lemma 6, an increase in the defendant's expected litigation loss,  $l_D$ , increases the spending on care and, therefore, reduces the probability of an accident.

(3) Totally differentiating equations (1) and (2) yields  $d\nu = dS \left[ C_D \frac{\partial g(S)}{\partial S} - g(S) \right]$  and  $(S - \tilde{A} - C_D)g(\tilde{A})d\tilde{A} = Sd\nu$ . From the last equation:  $\frac{\partial \nu}{\partial \tilde{A}} = \frac{(S - \tilde{A} - C_D)g(\tilde{A})}{S}$  and  $\frac{\partial S}{\partial \tilde{A}} = \frac{(\tilde{A} + C_D)g(\tilde{A})}{S \left[g(S) - C_D \frac{\partial g(S)}{\partial S}\right]}$ . By condition (5), the last expression is greater than zero.

(4) The probability of a zero offer is:  $\beta = 1 - \frac{f}{\gamma S}$ . Then,  $\frac{\partial \beta}{\partial \tilde{A}} = \frac{f}{\gamma S^2} \frac{\partial S}{\partial \tilde{A}} > 0$ .

(5) Let  $\zeta = \int_{\tilde{A}}^{\tilde{A}} g(A) dA + \nu = 1 - G(\tilde{A}) + \nu$  represent the aggregate filing. Then,  $\frac{\partial \zeta}{\partial \tilde{A}} = -g(\tilde{A}) + \frac{\partial \nu}{\partial \tilde{A}} = -\frac{(\tilde{A}+C_D)g(\tilde{A})}{C} < 0.$ 

(6) The probability of trial (conditional on accident occurrence) is:  $\rho = \beta(1-G(\tilde{A})) + (1-\beta)(1-G(S)) = 1 - G(\tilde{A}) - \frac{f(G(S)-G(\tilde{A}))}{\gamma S}$ . Then,  $\frac{\partial \rho}{\partial \tilde{A}} = -g(\tilde{A}) + \frac{\partial S}{\partial \tilde{A}} \frac{f}{\gamma} \frac{[G(S)-G(\tilde{A})-g(S)S]}{S^2} < 0$ . The last inequality holds because  $G(S) - G(\tilde{A}) - g(S)S = C_D g(S) - \nu - g(S)S = (C_D - S)g(S) - \nu < (C_D - \tilde{A})g(S) - \nu < 0$ , when  $C_D < \tilde{A}$ .

**Proposition 4.** A reduction in  $\hat{A}$  reduces No-Access to Justice  $\eta$ .

**Proof.** 
$$\eta = \left[\int_{0}^{\tilde{A}} g(A)dA - \nu\right] + \nu\beta$$
. Then,  $\frac{\partial\eta}{\partial\tilde{A}} = g(\tilde{A}) - \frac{\partial\nu}{\partial\tilde{A}} + \frac{\partial\nu}{\partial\tilde{A}}\beta + \frac{\partial\beta}{\partial\tilde{A}}\nu = g(\tilde{A}) - \left[\frac{(S-\tilde{A}-C_D)}{S}\right]g(\tilde{A})\frac{f}{\gamma S} + \frac{\partial\beta}{\partial\tilde{A}}\nu > g(\tilde{A}) - g(\tilde{A})\left[1 - \frac{(\tilde{A}+C_D)}{S}\right] + \frac{\partial\beta}{\partial\tilde{A}}\nu > 0.$ 

**Proposition 5.** There exists  $\bar{f} > 0$  such that, for any  $f < \bar{f}$ ,  $l_W - l_D > 0$ .

**Proof.** Equation (11) can be rewritten as:  $l_W - l_D = \left[\int_0^{\tilde{A}} Ag(A)dA - \int_{\tilde{A}}^{\tilde{A}} Ag(A)dA\right] + \left\{C_P[1 - G(\tilde{A})] - \frac{f(C_P + C_D)}{\gamma S}[G(S) - G(\tilde{A})]\right\} + \left\{\left[\int_{\tilde{A}}^{\tilde{A}} g(A)dA + \nu\right]f + \theta\left[\int_0^{\tilde{A}} g(A)dA - \nu + \nu\beta\right]\right\}$ . The first term (brackets) and the third term (curly brackets) are positive. We will show that the second term (curly brackets) is positive for sufficiently small values of f. Four steps are included. (1)  $\tilde{A} = \frac{C_P + (f + C_P - x)r}{\gamma}$  is a continuous function of f.

(2) We have already proved that  $\frac{\partial S}{\partial \tilde{A}} > 0$ . If  $S(\tilde{A})$  is a differentiable function, then it is also continuous in  $\tilde{A}$ .

(3)  $S(\tilde{A}(f))$  is a continuous function of f. Therefore,  $\Gamma(f) \equiv \left\{ C_P[1 - G(\tilde{A})] - \frac{f(C_P + C_D)}{\gamma S(\tilde{A}(f))} [G(S(\tilde{A}(f))) - G(\tilde{A})] \right\}$  is also a continuous function of f.

 $(4) \lim_{f \to 0} \Gamma(f) = \lim_{f \to 0} \left\{ C_P[1 - G(\tilde{A})] - \frac{f(C_P + C_D)}{\gamma S(\tilde{A}(f))} [G(S(\tilde{A}(f))) - G(\tilde{A})] \right\} = C_P[1 - G(\tilde{A})] > 0. \text{ Hence by continuity of } \Gamma(f) \text{, there exists } \bar{f} \text{, such that for any } f < \bar{f}, \left\{ C_P[1 - G(\tilde{A})] - \frac{f(C_P + C_D)}{\gamma S} [G(S) - G(\tilde{A})] \right\} > 0.$ 

#### **Proposition 6.** A cost-reducing policy is welfare improving if and only $\theta > \underline{\theta}$ .

**Proof.** Define  $\underline{\theta} \equiv \frac{-\Omega}{\Lambda}$ . Define  $\Theta(\theta) \equiv \Omega + \Lambda \theta$ . By assumption,  $\theta \geq 0$ . By propositions 3 and 4,  $\Lambda > 0$ . Hence,  $\Theta(\theta) = \Omega + \Lambda \theta$  is an upward-sloping linear function in  $\theta$ . There two possible cases. (1)  $\Omega > 0$ :  $\underline{\theta} < 0$ . Then,  $\theta > \underline{\theta}$  and  $\Theta(\theta) > 0$ . Hence,  $\Theta(\theta) > 0$  if and only if  $\theta > \underline{\theta}$ . (2)  $\Omega \leq 0$ :  $\underline{\theta} \geq 0$ . Then,  $\Theta(\underline{\theta}) = 0$ . Hence,  $\Theta(\theta) > 0$  if and only if  $\theta > \underline{\theta}$ .

# 2 Pure-Strategy PBE – Proofs

Assume that conditions (3), (4'), (5) and (6') hold.

**Lemma 8.** All cases are filed in equilibrium; the equilibrium settlement offer  $S \in (\tilde{A}, \bar{A})$ , implicitly defined by  $G(S) = C_D g(S)$ , exists and is unique.

**Proof.** First, it is simple to show that offers greater than  $\overline{A}$  are strictly dominated by an offer equal to  $\overline{A}$ . Similarly, offers  $S \in (0, \widetilde{A})$  are not in the set of equilibrium offers. These proofs follow the logic applied in Lemma 1. Second, we will demonstrate that a zero offer cannot be an equilibrium offer in pure strategy. Suppose not. The defendant always makes a zero offer in equilibrium. Then, only cases with types  $A \ge \widehat{A}$  will be filed. When facing a plaintiff of a type  $\widehat{A} \le A \le \widehat{A} + \epsilon$ , the defendant's expected litigation loss will be lower by offering  $\widehat{A} + \epsilon$ , where  $\epsilon > 0$  (small number):  $\int_{\widehat{A}}^{\widehat{A} + \epsilon} (\widehat{A} + \epsilon)g(A)dA < \int_{\widehat{A}}^{\widehat{A} + \epsilon} (A + C_D)g(A)dA$ , when  $C_D > \epsilon$ . Contradiction follows. Hence, a zero offer cannot be an equilibrium offer in pure strategy.

Third, we will show that a mixed-strategy with a zero offer and an offer  $S \in [\tilde{A}, \bar{A}]$  is not in the set of equilibrium offers.<sup>8</sup> Suppose not. The defendant mixes between a zero offer and a positive offer

 $<sup>^{8}</sup>$ S might be non-unique, i.e., multiple S might be made in equilibrium with positive probabilities.

 $S \in [\tilde{A}, \bar{A}]$  in equilibrium. By condition (3) and following the logic applied in the proof of Lemma 2, we can establish that all cases with  $A \ge \tilde{A}$  and some low-damage cases must be filed. Then, the defendant's expected litigation loss, which is the same under both offers, is:  $\nu S + \int_{\tilde{A}}^{S} Sg(A)dA + \int_{S}^{\tilde{A}} (A+C_D)g(A)dA = \int_{\tilde{A}}^{\tilde{A}} (A+C_D)g(A)dA$ . When the defendant offers a unique  $S_1 \in [\tilde{A}, \bar{A}]$ , all cases are filed. His expected litigation loss is  $G(S_1)S_1 + \int_{S_1}^{\tilde{A}} (A+C_D)g(A)dA$ . By condition (4'), there exists  $S_1 \in [\tilde{A}, \bar{A}]$  such that,  $G(S_1)S_1 < \int_{\tilde{A}}^{S_1} (A+C_D)g(A)dA$ . Then,  $G(S_1)S_1 + \int_{S_1}^{\tilde{A}} (A+C_D)g(A)dA < \int_{\tilde{A}}^{\tilde{A}} (A+C_D)g(A)dA$ , where the left- and right-hand sides represent the defendant's expected litigation loss when he makes an offer  $S_1 \in [\tilde{A}, \bar{A}]$  and when he mixed between a zero offer and a positive offer  $S \in [\tilde{A}, \bar{A}]$ , respectively. The defendant is better off, and then, will always deviate to offer  $S_1 \in [\tilde{A}, \bar{A}]$ . Contradiction follows. Hence, a mixed strategy with a zero offer and a positive offer  $S \in [\tilde{A}, \bar{A}]$  cannot be in set of equilibrium offers.

Fourth, from the previous analysis, we conclude that the set of equilibrium offers might only involve a pure strategy with an offer  $S \in [\tilde{A}, \bar{A}]$  or a mixed strategy with multiple offers  $S \in [\tilde{A}, \bar{A}]$ . By condition (3), lawyers with high- and low-damage cases will get positive expected payoffs. Then, all cases will be filed. Hence,  $\nu = G(\tilde{A})$  and the total filed cases  $\zeta = 1$ , in equilibrium.

Fifth, we will show that the equilibrium offer  $S \in [\bar{A}, \bar{A}]$  exists and is unique. Under both, the pure and mixed strategies, the defendant's expected litigation loss  $l_D(S) = G(S)S + \int_S^{\bar{A}} (A + C_D)g(A)dA$ , by condition (1). Minimization of  $l_D(S)$  yields the first-order condition:  $G(S) = C_Dg(S)$ . By condition (5), the second derivative  $g(S) - C_D \frac{\partial g(S)}{\partial S} > 0$ , for all  $S \in [\tilde{A}, \bar{A}]$ . Then, the function is strictly convex, and has a unique minimum. Hence, the set of equilibrium offers must involve a unique offer  $S \in [\tilde{A}, \bar{A}]$ . Fifth, we will demonstrate that the equilibrium offer is interior,  $S \in (\tilde{A}, \bar{A})$ . There are three possible options for a unique minimum of the function  $l_D(S) = G(S)S + \int_S^{\tilde{A}} (A + C_D)g(A)dA$ . (1) The function  $l_D(S)$  is strictly increasing on the interval  $[\tilde{A}, \bar{A}]$ ; it achieves a unique corner minimum at  $S = \tilde{A}$ . (2) The function  $l_D(S)$  is strictly decreasing on the interval  $[\tilde{A}, \bar{A}]$ ; it achieves a corner minimum at  $S = \tilde{A}$ . (3) The function  $l_D(S)$  achieves a unique interior minimum on the interval  $[\tilde{A}, A]$ ; the value of S is implicitly defined by the first-order condition. Consider option (1). When  $S = \tilde{A}$ , all cases are filed, by condition (3). Then, the defendant is better off by making a zero offer:  $G(\tilde{A})\tilde{A} + \int_{\tilde{A}}^{\tilde{A}} (A + C_D)g(A)dA > \int_{\tilde{A}}^{\tilde{A}} (A + C_D)g(A)dA$ , where the right-hand side term corresponds to the defendant's expected litigation loss from making a zero offer. Contradiction follows. Hence, an offer  $S = \tilde{A}$  cannot be an equilibrium offer. Consider options (2) and (3). The necessary and sufficient conditions for options (2) and (3) to occur are  $\lim_{S \to \tilde{A}^-} \frac{\partial l_D(S)}{\partial S} = 1 - C_D g(\bar{A}) \leq 0$ , and  $\lim_{S \to \tilde{A}^-} \frac{\partial l_D(S)}{\partial S} = 1 - C_D g(\bar{A}) > 0$ . Hence, the function has an interior minimum  $S \in (\tilde{A}, \bar{A})$ .

**Lemma 9.** The defendant's equilibrium probability of an accident  $\lambda_D = \arg \min K(\lambda) + \lambda l_D$  exists, is unique, and is decreasing in  $l_D$ .

**Proof.** Apply the logic used in the proof of Lemma 6.  $\blacksquare$ 

Lemma 10. In equilibrium, the defendant's prior and posterior beliefs are the same.

**Proof.** Given that all cases are filed, the defendant cannot update his beliefs upon observing filing. Hence, his posterior and prior beliefs are the same. The defendant's equilibrium beliefs are as follows. For any  $y \in (\tilde{A}, \bar{A}]$ ,  $P(0 < A \leq y | Filing) = G(y)$ .

# [TO BE AVAILABLE ONLINE ONLY]

# FINANCIALLY-CONSTRAINED LAWYERS: AN ECONOMIC THEORY OF LEGAL DISPUTES

# SUPPLEMENTARY MATERIAL

Claudia M. Landeo and Maxim Nikitin

November 14, 2017

# 1 Introduction

This document presents supplementary material. Section 2 provides formal analysis of the additional equilibria of the general model: Mixed- and pure-strategy PBE with corner solution (i.e., where the equilibrium positive settlement offer is equal to  $\bar{A}$ , the maximum possible damage level). Section 3 provides formal analysis of a version of our model under a uniform distribution of damages and presents the MATLAB program used to compute the numerical examples of deterrence and social welfare discussed in the paper. Note that some conditions and propositions used in the general model directly apply to this document. If a condition or a proposition is similar to a condition presented in the general model, then the letter "A" or the letter "B" (Sections 2 and 3, respectively) is placed besides the number.

# 2 Mixed- and Pure-Strategy PBE with Corner Solution

#### 2.1 Mixed-Strategy PBE with Corner Solution

7

Proposition 1A characterizes the mixed-strategy PBE with a corner solution under conditions (3)-(5) and (6A).

$$f < \gamma \tilde{A}.$$
 (3)

$$\min_{\tau \in [\tilde{A}, \bar{A}]} \left\{ G(\tau)\tau - \int_{\tilde{A}}^{\tau} (A + C_D)g(A)dA \right\} > 0.$$

$$\tag{4}$$

For any  $A \in [\tilde{A}, \bar{A}]$ ,

$$g(A) - C_D \frac{\partial g(A)}{\partial A} > 0.$$
(5)

$$\int_{\tilde{A}}^{A} Ag(A)dA + C_{D}[1 - G(\tilde{A}) - \bar{A}g(\bar{A})] \le 0.$$
(6A)

**Proposition 1A.** Assume condition (3)-(5) and (6A) hold. The following strategy profile, together with the defendant's beliefs, characterize the mixed-strategy perfect Bayesian equilibrium with a corner solution.

(1) The defendant chooses a probability of accident  $\lambda_D = \arg \min \{K(\lambda) + \lambda \int_{\bar{A}}^{\bar{A}} (A + C_D)g(A)dA\}$ . If a lawsuit is filed, the defendant mixes between proposing a zero offer with probability  $\beta = 1 - \frac{f}{\gamma S}$  and proposing an offer  $S = \bar{A}$  with the complementary probability.

(2) A high-damage case is always filed by the plaintiff's lawyer; an average low-damage case is filed with probability  $\frac{\nu}{G(\tilde{A})}$ .

(3) A high-damage plaintiff always rejects a zero offer and accepts an offer  $S = \overline{A}$  only if  $A \leq \overline{A}$ ; a low-damage plaintiff always accepts an offer  $S = \overline{A}$ .

(4) The defendant's equilibrium beliefs are as follows. When the defendant observes a lawsuit, he believes that  $P(0 < A < \tilde{A}) = \frac{\nu}{\nu+1-G(\tilde{A})}, \ P(\tilde{A} \le A \le y) = \frac{G(y)-G(\tilde{A})}{\nu+1-G(\tilde{A})} \text{ for any } y \in [\tilde{A}, \bar{A}].$ 

**Proof.** The proof consists of several steps.

Step 1. We will first demonstrate that that offers greater than  $\overline{A}$  and offer  $S \in (0, \widetilde{A})$  are not in the set of equilibrium offers. Second, we will show that a zero offer must be in the set of equilibrium offers. Third, we will demonstrate that a zero offer cannot be the only equilibrium offer: At least one strictly positive offer  $S \in [\widetilde{A}, \overline{A}]$  must be in the set of equilibrium offers, in addition to a zero offer. The proofs follow the logic applied in the proof of Lemma 1.

Step 2. We will show that there must be some low-damage cases that are filed ( $\nu > 0$ ), and that all high-damage cases are filed. The proofs follow the logic applied in the proof of Lemma 2.

Step 3. We will demonstrate that the strictly positive equilibrium offer is A. By condition (6A), the system (1)-(2) does not have an interior solution. By condition (5), the second derivative of  $l_D(S)$ ,  $g(S) - C_D \frac{\partial g(S)}{\partial S} > 0$ , which is satisfied for all  $S \in [\tilde{A}, \bar{A}]$ . Then, there is a unique minimum. There are two potential options. (1)  $l_D(S)$  is strictly increasing and the optimal strictly positive offer is  $\tilde{A}$ ; (2)  $l_D(S)$  is strictly decreasing and the optimal strictly positive offer is  $\tilde{A}$ . Consider Option 1. Suppose that Option 1 holds. The defendant is indifferent between a zero offer and  $\tilde{A}$ . The indifference condition implies:  $\int_{\tilde{A}}^{\tilde{A}} (A+C_D)g(A)dA = \nu \tilde{A} + \int_{\tilde{A}}^{\tilde{A}} (A+C_D)g(A)dA$ . This condition holds only if  $\nu = 0$ . Contradiction follows. Consider Option 2 now. The defendant is indifferent between a zero offer and  $\tilde{A}$ . The indifference condition implies:  $[\nu + 1 - G(\tilde{A})]\bar{A} = \int_{\tilde{A}}^{\tilde{A}} (A+C_D)g(A)dA$ . Solving for  $\nu$  yields:  $\nu = \frac{\int_{\tilde{A}}^{\tilde{A}} (A+C_D)g(A)dA}{A} - [1 - G(\tilde{A})]$ . By Claim 2 in the Appendix,  $\nu < G(\tilde{A})$ . Hence,  $S = \bar{A}$  and  $\nu = \frac{\int_{\tilde{A}}^{\tilde{A}} (A+C_D)g(A)dA}{A} - [1 - G(\tilde{A})]$  are the equilibrium strictly positive offer and the equilibrium mass of low-damage cases that are filed.

Step 4. The composition of  $\nu$  determines the equilibrium probability that the defendant makes a zero offer  $\beta$  The proof follows the logic applied in the proof of Lemma 5.

Step 5. The defendant's optimal probability of an accident  $\lambda_D = \arg \min\{K(\lambda) + \lambda l_D\} = \arg \min\{K(\lambda) + \lambda \int_{\bar{A}}^{\bar{A}} (A + C_D)g(A)dA\}$ . The proof follows the logic applied in the proof of Lemma 6.

Step 6. The equilibrium strategies of the average plaintiff and his lawyer and the equilibrium mass of filed cases determine the beliefs of the defendant. The proof follows the logic applied in Lemma 7.  $\blacksquare$ 

#### 2.2 Pure-Strategy PBE with Corner Solution

Proposition 2A characterizes the pure-strategy PBE with a corner solution under conditions (3), (4'), (5), and (6'A).

$$f < \gamma \tilde{A}.$$
 (3)

$$\min_{\tau \in [\tilde{A}, \bar{A}]} \left\{ G(\tau)\tau - \int_{\tilde{A}}^{\tau} (A + C_D)g(A)dA \right\} < 0.$$

$$(4')$$

For any  $A \in [\tilde{A}, \bar{A}]$ ,

$$g(A) - C_D \frac{\partial g(A)}{\partial A} > 0.$$
(5)

$$1 - C_D g(\bar{A}) \le 0. \tag{6'A}$$

**Proposition 2A.** Assume conditions (3), (4'), (5), and (6'A) hold. The following strategy profile, together with the defendant's beliefs, characterize the pure-strategy perfect Bayesian equilibrium with a corner solution.

(1) The defendant chooses a probability of accident  $\lambda_D = \arg \min \{K(\lambda) + \lambda l_D(S)\}$ , where  $l_D(S) = G(S)S + \int_S^{\bar{A}} (A + C_D)g(A)dA$ . If a lawsuit is filed, the defendant proposes an offer  $S = \bar{A}$  with certainty. (2) All cases are filed by the plaintiff's lawyer.

(3) A plaintiff always accepts an offer  $S = \overline{A}$ .

(4) The defendant's equilibrium beliefs are as follows. When the defendant observes a lawsuit, he believes that  $P(0 < A \le y) = G(y)$ .

**Proof.** The proof consists of several steps.

Step 1. We will first demonstrate that offers greater than  $\overline{A}$  and offer  $S \in (0, \widehat{A})$  are not in the set of equilibrium offers. Second, we will show that a zero offer cannot be an equilibrium offer. Third, we will demonstrate that a mixed-strategy with a zero offer and  $S \in [\widetilde{A}, \widehat{A}]$  cannot be in the set of equilibrium offers. The proofs follow the logic applied in the proof of Lemma 1.

Step 2. From the previous analysis, we conclude that the set of equilibrium offers might only involve a pure strategy with a unique offer  $S \in [\tilde{A}, \bar{A}]$  or a mixed strategy with multiple offers  $S \in [\tilde{A}, \bar{A}]$ . By condition (3), lawyers with high- and low-damage cases will get positive expected payoffs. Then, all cases will be filed. Hence,  $\nu = G(\tilde{A})$  and the total filed cases  $\zeta = 1$ , in equilibrium.

Step 3. We will first show that the equilibrium offer  $S \in [\tilde{A}, \bar{A}]$  exists and is unique. Under both, the pure and the mixed strategies, the defendant's expected litigation loss  $l_D(S) = G(S)S + \int_S^{\tilde{A}} (A + C_D)g(A)dA$ , by equation (1). Minimization of  $l_D(S)$  yields the first-order condition:  $G(S) = C_Dg(S)$ . By condition (5), the second derivative  $g(S) - C_D \frac{\partial g(S)}{\partial S} > 0$ , for all  $S \in [\tilde{A}, \bar{A}]$ . Then, the function is strictly convex and has a unique minimum. Hence, the set of equilibrium offers must involve a unique offer  $S \in [\tilde{A}, \bar{A}]$ . Second, we will demonstrate that the equilibrium offer is corner,  $S = \bar{A}$ . There are three possible options for a unique minimum of the function  $l_D(S)$ . (1) The function  $l_D(S)$  is strictly increasing on the interval  $[\tilde{A}, \bar{A}]$ ; it achieves a unique corner minimum at  $S = \tilde{A}$ . (2) The function  $l_D(S)$  is strictly decreasing on the interval  $[\tilde{A}, \bar{A}]$ ; it achieves a corner minimum at  $S = \bar{A}$ . (3) The function  $l_D(S)$  achieves a unique interior minimum on the interval  $(\tilde{A}, \bar{A})$ ; the value of S is implicitly defined by the first-order condition. Consider option (1). When  $S = \tilde{A}$ , all cases are filed, by condition (3). Then, the defendant is better off by making a zero offer:  $G(\tilde{A})\tilde{A} + \int_{\tilde{A}}^{\tilde{A}}(A + C_D)g(A)dA > \int_{\tilde{A}}^{\tilde{A}}(A + C_D)g(A)dA$ , where the right-hand side term corresponds to the defendant's expected litigation loss from making a zero offer. Contradiction follows. Hence, an offer  $S = \tilde{A}$  cannot be an equilibrium offer. Consider options (2) and (3). The necessary and sufficient conditions for options (2) and (3) to occur are  $\lim_{S \to \tilde{A}^-} \frac{\partial l_D(S)}{\partial S} = 1 - C_Dg(\bar{A}) \leq 0$ , and  $\lim_{S \to \tilde{A}^-} \frac{\partial l_D(S)}{\partial S} = 1 - C_Dg(\bar{A}) > 0$ , respectively. By condition (6'A),  $\lim_{S \to \tilde{A}^-} \frac{\partial l_D(S)}{\partial S} = 1 - C_Dg(\bar{A}) \leq 0$ . Hence, the function has a corner minimum at  $S = \bar{A}$ .

Step 4. The defendant's optimal probability of an accident  $\lambda_D = \arg \min \{K(\lambda) + \lambda l_D(S)\}$ . The proof follows the logic applied in the proof of Lemma 6.

Step 5. We will show that, in equilibrium, the defendant's posterior and prior beliefs are the same. The proof follows the logic applied in the proof of Lemma 10.  $\blacksquare$ 

# 3 A Model with Uniform Distribution of Damages

This section refers to a version of our model under a uniform distribution of damages. We first present formal analysis of the mixed- and pure-strategy PBE. We then present the MATLAB program related to the numerical analysis of deterrence and the numerical analysis of the effect of a change in r on social welfare, which are discussed in the main text of the paper.

Assume A is uniformly distributed on the interval  $(0, \overline{A}]$ . Then,  $G(A) = \frac{A}{\overline{A}}$  and  $g(A) = \frac{1}{\overline{A}}$ . Assume also that  $K(\lambda)$  has the following functional form:  $K(\lambda) = B\left[1 - \sqrt{1 - (\lambda - 1)^2}\right]$ . This function satisfies the assumptions stated in the paper:  $K'(\lambda) = -\frac{B(1-\lambda)}{\sqrt{1-(\lambda-1)^2}} < 0$ ,  $K''(\lambda) = B\left[1 - (\lambda - 1)^2\right]^{-3/2} > 0$ ,  $\lim_{\lambda \to 0+} K'(\lambda) = +\infty$ ,  $\lim_{\lambda \to 1-} K'(\lambda) = 0$ . The other assumptions of our general model hold.

#### 3.1 Mixed-Strategy PBE

## **Technical Conditions**

$$f < \gamma \tilde{A}.$$
(3)

$$\min_{\tau \in [\tilde{A}, \bar{A}]} \{ (\tau - C_D)^2 + \tilde{A}^2 + 2\tilde{A}C_D - C_D^2 \} > 0.$$
(4B)

$$\frac{\bar{A}^2}{2} - \tilde{A}^2/2 - C_D \tilde{A} > 0.$$
(6B)

Condition 5 holds trivially under a uniform distribution of damages. Condition (4B) implies  $\tilde{A} > C_D(\sqrt{2}-1)$ . Intuitively, the mixed-strategy PBE arises in a state of the world is characterized by a high level of lawyers' constraints.

#### **Equilibrium Strategies and Outcomes**

Equilibrium S and  $\nu$  are jointly determined by the system of equations (1B)-(2B):

$$\nu S + \int_{\tilde{A}}^{S} S \frac{1}{\bar{A}} dA + \int_{S}^{\bar{A}} (A + C_D) \frac{1}{\bar{A}} dA = \int_{\tilde{A}}^{\bar{A}} (A + C_D) \frac{1}{\bar{A}} dA \tag{1B}$$

$$\nu + \frac{S}{\bar{A}} - \frac{\tilde{A}}{\bar{A}} = \frac{C_D}{\bar{A}}.$$
(2B)

The system of equations (1B)-(2B) has an explicit solution. The equilibrium positive offer S and the equilibrium mass of low-damage cases that are filed  $\nu$  are:

$$S = \sqrt{\tilde{A}^2 + 2\tilde{A}C_D}$$

and

$$\nu = \frac{\tilde{A} + C_D - S}{\bar{A}} = \frac{\tilde{A} + C_D - \sqrt{\tilde{A}^2 + 2\tilde{A}C_D}}{\bar{A}}.$$

The additional equilibrium outcomes are as follows. Mass of filed cases:  $\zeta = 1 - G(\tilde{A}) + \nu = 1 + \left(\frac{C_D - S}{\tilde{A}}\right)$ . Probability of a zero offer:  $\beta = 1 - \frac{f}{\gamma S}$ . Mass of cases that proceed to trial:  $\rho = 1 - \frac{\tilde{A}}{\tilde{A}} - \frac{f}{\gamma \tilde{A}} + \frac{f \tilde{A}}{\gamma \tilde{A}S}$ . Defendant's expected litigation loss:  $l_D = \frac{0.5 \bar{A}^2 + C_D \bar{A} - \frac{\tilde{A}^2}{2} - C_D \tilde{A}}{\bar{A}}$ . No-access-to-justice component:  $\eta = \frac{\tilde{A}}{\tilde{A}} + \frac{f}{\gamma \tilde{A}} - \left(\frac{\tilde{A} + C_D}{\tilde{A}}\right) \frac{f}{\gamma S}$ . Probability of accident:  $\lambda_D = 1 - \frac{l_D}{\bar{A}}$ .

$$\lambda_D = 1 - \frac{l_D}{\sqrt{B^2 + l_D^2}}.$$

# Social Welfare and Deterrence

#### Social Welfare Loss Function

$$SWL = K(\lambda) + \lambda l_W,$$

where

$$l_W = H + \zeta f + \rho (C_P + C_D) + \theta \eta =$$
  
=  $\frac{\bar{A}}{2} + \left(1 + \frac{C_D}{\bar{A}} - \frac{S}{\bar{A}}\right) f + \left(1 - \frac{\tilde{A}}{\bar{A}} - \frac{f}{\gamma \bar{A}} + \frac{f \tilde{A}}{\gamma \bar{A}S}\right) (C_P + C_D) + \theta \left(\frac{\tilde{A}}{\bar{A}} + \frac{f}{\gamma \bar{A}} - \frac{f(\tilde{A} + C_D)}{\gamma \bar{A}S}\right).$ 

#### **Deterrence Components**

The defendant's expected litigation loss  $l_D$ , given by each side of equation (1B), can be expressed as:

$$\begin{split} l_D &= (1-\beta) \left[ \nu S + \int_{\tilde{A}}^{S} S \frac{1}{\bar{A}} dA + \int_{S}^{\bar{A}} (A+C_D) \frac{1}{\bar{A}} dA ) \right] + \\ &+ \beta \left[ \int_{\tilde{A}}^{\bar{A}} (A+C_D) \frac{1}{\bar{A}} dA \right] = \\ (1-\beta) \left[ \frac{C_D S}{\bar{A}} + \frac{1}{\bar{A}} \left( \frac{\bar{A}^2}{2} + C_D \bar{A} - \frac{S^2}{2} - C_D S \right) \right] + \frac{\beta}{\bar{A}} \left[ \frac{\bar{A}^2}{2} + C_D \bar{A} - \frac{\tilde{A}^2}{2} - C_D \tilde{A} \right]. \end{split}$$
 is given by

Then,  $l_W - l_D$  is given by:

=

$$l_W - l_D =$$

$$= \left\{ \left(1 + \frac{C_D - S}{\bar{A}}\right) f + \left[\beta \left(1 - \frac{\tilde{A}}{\bar{A}}\right) + (1 - \beta) \left(1 - \frac{S}{\bar{A}}\right)\right] C_P + \theta \left(\frac{\tilde{A}}{\bar{A}} - \nu + \nu\beta\right) \right\} + \left\{ H - \left\{ (1 - \beta) \left[\frac{C_D S}{\bar{A}} + \frac{1}{\bar{A}} \left(\frac{\bar{A}^2}{2} - \frac{S^2}{2}\right)\right] + \frac{\beta}{\bar{A}} \left[\frac{\bar{A}^2}{2} - \frac{\tilde{A}^2}{2}\right] \right\} \right\}.$$

Effects of a Cost-Reducing Policy on Social Welfare

Direct Effect (DE)

$$\frac{\partial SWL(\lambda_D)}{\partial \tilde{A}} = \lambda_D f \frac{\partial \zeta}{\partial \tilde{A}} + \lambda_D (C_P + C_D) \frac{\partial \rho}{\partial \tilde{A}} + \lambda_D \theta \frac{\partial \eta}{\partial \tilde{A}} = \\ = \lambda_D \left( f \frac{\partial \zeta}{\partial \tilde{A}} + (C_P + C_D) \frac{\partial \rho}{\partial \tilde{A}} + \theta \frac{\partial \eta}{\partial \tilde{A}} \right) =$$

$$=\lambda_D \left\{ f \frac{\tilde{A} + C_D}{\bar{A}S} + (C_P + C_D) \frac{1}{\bar{A}} \left( 1 - \frac{f \tilde{A} C_D}{\gamma S^3} \right) + \theta \left( \frac{1}{\bar{A}} + \frac{f C_D^2}{\gamma \bar{A} S^3} \right) \right\}.$$

Indirect Effect (IE)

$$\frac{\partial SWL(\lambda_D)}{\partial \lambda_D} \frac{\partial \lambda_D}{\partial \tilde{A}} = (l_W - l_D) \frac{\partial \lambda_D}{\partial \tilde{A}} = \\ = \left\{ \left( \zeta f + \rho C_P + \theta \eta \right) + \right. \\ \left. + \left\{ H - \left\{ (1 - \beta) \left[ \frac{C_D S}{\bar{A}} + \frac{1}{\bar{A}} \left( \frac{\bar{A}^2}{2} - \frac{S^2}{2} \right) \right] + \frac{\beta}{\bar{A}} \left[ \frac{\bar{A}^2}{2} - \frac{\tilde{A}^2}{2} \right] \right\} \right\} \right\} \left\{ \frac{(\tilde{A} + C_D) [1 - (1 - \lambda_D)^2]^{3/2}}{\bar{A}B} \right\}.$$

#### **Overall Effect (OE)**

The overall effect of a cost-reducing policy is:

$$\frac{dSWL(\lambda_D)}{d\tilde{A}} = \left[ (l_W - l_D) \frac{\partial \lambda_D}{\partial \tilde{A}} \right] + \left[ \lambda_D f \frac{\partial \zeta}{\partial \tilde{A}} + \lambda_D (C_P + C_D) \frac{\partial \rho}{\partial \tilde{A}} + \lambda_D \theta \frac{\partial \eta}{\partial \tilde{A}} \right].$$

The overall effect can be written as:

$$\frac{dSWL(\lambda_D)}{d\tilde{A}} = \Omega + \Lambda\theta$$

where:

$$\Omega = \left\{ \left(1 + \frac{C_D - S}{\bar{A}}\right) f + \left[\beta\left(1 - \frac{\tilde{A}}{\bar{A}}\right) + (1 - \beta)\left(1 - \frac{S}{\bar{A}}\right)\right] C_P \right\} + \left\{H - \left\{(1 - \beta)\left[\frac{C_D S}{\bar{A}} + \frac{1}{\bar{A}}\left(\frac{\bar{A}^2}{2} - \frac{S^2}{2}\right)\right] + \frac{\beta}{\bar{A}}\left[\frac{\bar{A}^2}{2} - \frac{\tilde{A}^2}{2}\right]\right\} \right\} \left\{\frac{(\tilde{A} + C_D)[1 - (1 - \lambda_D)^2]^{3/2}}{\bar{A}B}\right\} + \lambda_D \left\{-\frac{f}{\bar{A}}\left[\frac{(\tilde{A} + C_D)}{S}\right] + (C_P + C_D)\frac{1}{\bar{A}}\left[1 - \frac{f\tilde{A}C_D}{\gamma S^3}\right]\right\}$$

and

$$\Lambda = \left(\frac{\tilde{A}}{\bar{A}} - \nu + \nu\beta\right) \left\{ \frac{(\tilde{A} + C_D)[1 - (1 - \lambda_D)^2]^{3/2}}{\bar{A}B} \right\} + \lambda_D \left[\frac{1}{\bar{A}} + \frac{fC_D^2}{\gamma\bar{A}S^3}\right]$$

The threshold  $\underline{\theta}$  is defined as follows:  $\underline{\theta} \equiv \frac{-\Omega}{\Lambda}$ .

#### 3.2 Pure-Strategy PBE

# **Technical Conditions**

$$f < \gamma \tilde{A} \tag{3}$$

$$\min_{\tau \in [\tilde{A}, \bar{A}]} \{ (\tau - C_D)^2 + \tilde{A}^2 + 2\tilde{A}C_D - C_D^2 \} < 0$$
(4'B)

$$\bar{A} > C_D. \tag{6'B}$$

Condition 5 holds trivially under a uniform distribution of damages. Condition (4'B) implies  $\tilde{A} < C_D(\sqrt{2}-1)$ . Intuitively, the pure-strategy PBE arises in a state of the world characterized by a low level of lawyer's constraint.

#### **Equilibrium Strategies and Outcomes**

Equilibrium S:

$$\frac{S}{\bar{A}} = \frac{C_D}{\bar{A}}.\tag{8B}$$

Then,

$$S = C_D.$$

The additional equilibrium outcomes are as follows.

Mass of filed cases:  $\zeta = 1$ . Mass of low-damage cases that are filed:  $\nu = \frac{\tilde{A}}{\tilde{A}}$ . Mass of cases that proceed to trial:  $\rho = \frac{\tilde{A} - C_D}{\tilde{A}}$ . Defendant's expected litigation loss :  $l_D = C_D + 0.5\bar{A} - \frac{C_D^2}{2A}$ . Probability of accident:  $\lambda_D = 1 - \frac{l_D}{\sqrt{B^2 + l_D^2}}$ .

#### 3.3 Numerical Examples of Deterrence and Social Welfare

In Section 5.2 of the paper, we present a numerical example to illustrate the deterrence components. In Section 6.4 of the paper, we present a numerical example to illustrate the effect of a change in r (a specific costreducing policy) on social welfare. We use the uniform-distribution model presented in Section 3.3. The parameter set adopted is as follows:  $\{C_P, C_D, \gamma, f, x, \theta, \overline{A}, B\} = \{200, 500, .33, 70, 100, 750, 1200, 1000\}$ . One value of r, r = .18, is studied in the numerical example of deterrence. Three values of r are studied in the numerical example of social welfare:  $r \in \{.08, .18, .28\}$ .<sup>1</sup> The MATLAB software is used to construct the numerical examples. Figure 1 presents the MATLAB program.

<sup>&</sup>lt;sup>1</sup>The model conditions and the condition  $\theta > \underline{\theta}$  hold under the chosen exogenous parameters. In this numerical example, SWL is a monotonically increasing function of r. The value of r that minimizes SWL is r = .08. The values for  $\underline{\theta}$  are  $\underline{\theta} \in \{192, 158, 125\}$ , for  $r \in \{.08, .18, .28\}$ , respectively.

%Numerical Example on Deterrence and Numerical Example on the Effect of a Change in r on Social Wlefare

```
%Exogenous variables
```

```
Cp=200; Cd=500; f=70; x = 100; gamma = 0.33; Abar = 1200; B = 1000; theta=750;
for i = 1:1:3
 r(i) = -0.02+ i*0.10;
  %Conditions of the model
  Atilde(i)=(Cp+(Cp+f-x)*r(i))/gamma;
  if Cd < (sqrt(2)+1)*Atilde(i)
    if f < gamma*Atilde(i)
      if Cd - (Abar^2 - Atilde(i)^2)/(2*Atilde(i)) < 0
       %Computation of endogenous variables
        S(i) = sqrt((Atilde(i))^2 + 2*Atilde(i)*Cd);
        beta(i)=1- f/(gamma*S(i));
        nu(i) = (Atilde(i)+Cd-S(i))/Abar;
        rho(i) = beta(i)*(1 - Atilde(i)/Abar)+(1-beta(i))*(1-S(i)/Abar);
        eta(i) = Atilde(i)/Abar + f/(gamma*Abar) - (f*(Atilde(i)+Cd))/(gamma*S(i)*Abar);
        zeta(i) = 1+Cd/Abar - S(i)/Abar;
        probtrial(i) = rho(i)/zeta(i);
        ld(i) = (0.5*Abar^2 + Cd*Abar - 0.5*(Atilde(i))^2 - Cd*Atilde(i))/Abar;
        lambdad(i) = 1 - ld(i)/sqrt(B^2+(ld(i))^2);
        lw(i) = Abar/2 + zeta(i)*f + rho(i)*(Cp+Cd) + theta*eta(i);
        lambdaw(i) = 1 - lw(i)/sqrt(B^2+(lw(i))^2);
        SWL(i) = B^{*}(1-B/sqrt(B^{2} + (ld(i))^{2})) + lambdad(i)^{*}lw(i);
        deterrence(i) = lw(i)-ld(i);
        zeta_f(i) = zeta(i)*f;
        rho_Cp(i) = rho(i)*Cp;
        theta_eta(i) = theta*eta(i);
        expected_compensation_def(i) = (1-beta(i))*(nu(i) + S(i)/Abar - Atilde(i)/Abar)*S(i) +
        (beta(i)/Abar)*(Abar^2/2- (Atilde(i))^2/2) +((1-beta(i))/Abar)*(Abar^2/2 - (S(i))^2/2);
        H = Abar/2;
        derivative_rho_Atilde(i) = 1/Abar*(-1+(f*Atilde(i)*Cd)/(gamma*(S(i))^3));
        derivative_zeta_Atilde(i) = -(Atilde(i)+Cd)/(Abar*S(i));
derivative_eta_Atilde(i) = 1/Abar*(1+(f*Cd^2)/(gamma*(S(i))^3));
        K_prime_prime(i) = B*(1-(1-lambdad(i))^2)^(-1.5);
        Omega(i)= (0.5*Abar+ zeta(i)*f + rho(i)*(Cd+Cp)-
        ld(i))*(Atilde(i)+Cd)/(Abar*K_prime_prime(i))+lambdad(i)*((Cp+Cd)*derivative_rho_Atilde(i)-
        (Atilde(i)+Cd)*f/(Abar*S(i)));
        Lambda(i) = eta(i) *(Atilde(i)+Cd)/(Abar*K_prime_prime(i))+lambdad(i)*derivative_eta_Atilde(i);
        theta_under(i) = - Omega(i)/Lambda(i);
      end
    end
  end
end
```

%Computation of percentage changes when r decreases

```
for i = 2:1:3
    delta_zeta(i-1) = - 100*(zeta(i)-zeta(i-1))/zeta(i);
    delta_probtrial(i-1) = -100*(probtrial(i)-probtrial(i-1))/probtrial(i);
    delta_eta(i-1) = - 100*(eta(i)-eta(i-1))/eta(i);
    delta_nu(i-1) = - 100*(nu(i)-nu(i-1))/nu(i);
    delta_beta(i-1) = - 100*(beta(i)-beta(i-1))/beta(i);
    delta_ld(i-1) = -100*(ld(i)-ld(i-1))/ld(i);
    delta_lambda(i-1) = -100*(lambdad(i)-lambdad(i-1))/lambdad(i);
    delta_SWL(i-1) = - 100*(SWL(i)-SWL(i-1))/SWL(i);
    end
```

Figure 1: MATLAB Program

# Department of Economics, University of Alberta Working Paper Series

**2018-02:** Information and Transparency in Wholesale Electricity Markets: Evidence from Alberta – **Brown, D., Eckert, A.,** Lin, J.

**2018-01:** Does a Discount Rate Rule Ensure a Pension Plan Can Pay Promised Benefits without Excessive Asset Accumulation? – **Landon**, **S**., **Smith**, **C**.

**2017-13:** Income Inequality and Violence Against Women: Evidence from India – Rashad, A., **Sharaf, M.** 

**2017-12:** The Local Effects of the Texas Shale Boom on Schools, Students, and Teachers – **Marchand**, **J.**, Weber, J.

**2017-11:** Self-Sabotage in the Procurement of Distributed Energy Resources – **Brown**, **D**., Sappington, D.

2017-10: Public Private Competition – Klumpp, T., Su, X.

2017-09: Testing for State-Dependent Predictive Ability – Fossati, S.

**2017-08:** Default Risk, Productivity, and the Environment: Theory and Evidence from U.S. Manufacturing – **Andersen, D.** 

**2017-07:** Does Maternal Employment Affect Child Nutrition Status? New Evidence from Egypt – Rashad, A., **Sharaf, M.** 

**2017-06**: The Effect of Default Rates on Retail Competition and Pricing Decisions of Competitive Retailers: The Case of Alberta – **Brown**, **D.**, **Eckert**, **A**.

2017-05: Optimal Procurement of Distributed Energy Resources – Brown, D., Sappington, D.

**2017-04:** The Impact of Schooling Intensity on Student Learning: Evidence from a Quasi-Experiment – Andrietti, V., **Su**, **X**.

**2017-03:** The Voting Rights of Ex-Felons and Election Outcomes in the United States – **Klumpp**, **T.**, Mialon, H., Williams, M.

2017-02: Does the Design of a Fiscal Rule Matter for Welfare? – Landon, S., Smith, C.

**2017-01:** Carbon Pricing with an Output Subsidy under Imperfect Competition: The Case of Alberta's Restructured Electricity Market – **Brown**, **D**., **Eckert**, **A**., **Eckert**, **H**.

**2016-18:** Monetary Policy Tradeoffs Between Financial Stability and Price Stability – **Shukayev**, **M.**, Ueberfeldt, A.

**2016-17:** Managing Risk Taking with Interest Rate Policy and Macroprudential Regulations – Cociuba, S., **Shukayev, M.**, Ueberfeldt, A.

**2016-16:** On the Role of Maximum Demand Charges in the Presence of Distributed Generation Resources – **Brown**, **D.**, Sappington, D.

**2016-15:** Implementing Cross-Border Interbank Lending in BoC-GEM-FIN – **Shukayev**, **M.**, Toktamyssov, A.

**2016-14:** The Effects of Early Pregnancy on Education, Physical Health and Mental Distress: Evidence from Mexico – **Gunes**, **P.**, Tsaneva, M.

2016-13: An Equilibrium Selection Theory of Monopolization – Eckert, A., Klumpp, T., Su, X.

**2016-12:** Education Curriculum and Student Achievement: Theory and Evidence – Andrietti, V., **Su**, **X**.

2016-11: Poverty and Aging – Marchand, J., Smeeding, T.

**2016-10:** Local Labor Markets and Natural Resources: A Synthesis of the Literature – **Marchand**, **J.**, Weber, J.

**2016-09:** Accounting for Firm Exit and Loss of Variety in the Welfare Cost of Regulations – **Andersen**, **D**.